## ANNALS OF MATHEMATICS

# Local rigidity of partially hyperbolic actions I. KAM method and $\mathbb{Z}^k$ actions on the torus

By DANIJELA DAMJANOVIĆ and ANATOLE KATOK



SECOND SERIES, VOL. 172, NO. 3 November, 2010

ANMAAH

### Local rigidity of partially hyperbolic actions I. KAM method and $\mathbb{Z}^k$ actions on the torus

By DANIJELA DAMJANOVIĆ and ANATOLE KATOK

#### Abstract

We show  $C^{\infty}$  local rigidity for  $\mathbb{Z}^k$   $(k \ge 2)$  higher rank partially hyperbolic actions by toral automorphisms, using a generalization of the KAM (Kolmogorov-Arnold-Moser) iterative scheme. We also prove the existence of irreducible genuinely partially hyperbolic higher rank actions on any torus  $\mathbb{T}^N$  for any even  $N \ge 6$ .

#### 1. Introduction

1.1. Algebraic actions of higher rank abelian groups. Differentiable  $(C^r, r \ge 1)$  rigidity of the orbit structure and its variations sets the higher rank abelian group actions apart from the classical cases of diffeomorphisms and flows (actions of  $\mathbb{Z}$  and  $\mathbb{R}$ ) where only  $C^0$  orbit structure may be stable under small perturbations.

In this paper we begin the study of differentiable rigidity for algebraic (homogeneous of affine) *partially hyperbolic* actions of  $\mathbb{Z}^k \times \mathbb{R}^l$ ,  $k+l \ge 2$ . For definitions and general background on partially hyperbolic dynamical systems, see [32]. For a survey of local rigidity of actions of other groups, see [13]. In the latter area the key work dealing with the partially hyperbolic case is [14].

The most general condition in the setting of  $\mathbb{Z}^k \times \mathbb{R}^l$ ,  $k + l \ge 2$  actions, which leads to various rigidity phenomena (cocycle rigidity, local differentiable rigidity, measure rigidity, etc.), is the following:

( $\mathfrak{R}$ ) The group  $\mathbb{Z}^k \times \mathbb{R}^l$  contains a subgroup L isomorphic to  $\mathbb{Z}^2$  such that for the suspension of the restriction of the action to L every element other than identity acts ergodically with respect to the standard invariant measure obtained from Haar measure.

This condition should be viewed as a paradigm. We chose not to formulate specific conjectures due to various subtleties of algebraic nature which up to now have prevented even the results in the otherwise well-understood hyperbolic case from being clean and definitive; see [25]. Instead let us point out to two specific

The second named author was supported by NSF grants DMS-0071339 and DMS-0505539.

representative classes of algebraic actions where the above condition is satisfied and various (although not always the strongest expected) rigidity properties have been established in many cases.

(1) Actions of  $\mathbb{Z}^k$ ,  $k \ge 2$  by automorphisms or affine maps of tori or, more generally, (infra)nilmanifolds without nontrivial rank one factors for any finite cover, and suspensions of such actions.

(2) Left actions of a higher rank abelian group  $A \subset G$  on the double coset space  $C \setminus G/\Gamma$  where G is a connected Lie group,  $\Gamma \subset G$  a cocompact lattice, and C a compact subgroup of the centralizer of A which intersects with A trivially. Furthermore, the Lie algebras of A, C and contracting subgroups of all elements of A generate the full Lie algebra g of G.

1.2. Rigidity of actions and related notions. Let A be a finitely generated discrete group with generators  $g_1, \ldots, g_k$ . Let  $\operatorname{Act}^r(A, M)$  be the space of A actions by diffeomorphisms of class  $C^r$  of a compact manifold M. Every  $\alpha \in \operatorname{Act}^r(A, M)$  is defined by a homomorphism  $\rho_{\alpha}: A \to \operatorname{Diff}^r M$ . In particular, diffeomorphisms  $\rho_{\alpha}(g_1), \ldots, \rho_{\alpha}(g_k)$  completely determine the action  $\alpha$ . Thus the usual  $C^r$  topology of  $\operatorname{Diff}^r(M)^k$  induces via projection a topology in the space  $\operatorname{Act}^r(A, M)$  which we refer to as a  $C^r$  topology in the space of A actions on M. Similarly, for an action of a connected Lie group A on M, the  $C^r$  topology in  $\operatorname{Act}^r(A, M)$  is induced by the  $C^r$  topology on vector fields which generate the action of the Lie algebra of A.

For actions of discrete groups the notions of rigidity which we consider are summarized as follows.

An action  $\alpha$  of a finitely generated discrete group A on a manifold M is  $C^{k,r,l}$ locally rigid if any sufficiently  $C^r$  small  $C^k$  perturbation  $\tilde{\alpha}$  is  $C^l$  conjugate to  $\alpha$ ; i.e., there exists a  $C^l$  close to identity diffeomorphism  $\mathcal{H}$  of M which conjugates  $\tilde{\alpha}$  to  $\alpha$ :

(1.1) 
$$\mathcal{H} \circ \alpha(g) = \widetilde{\alpha}(g) \circ \mathcal{H}$$

for all  $g \in A$ .  $C^{\infty,1,\infty}$  local rigidity is often referred to as  $C^{\infty}$  local rigidity. The case of  $C^{1,1,0}$  is known as  $C^1$  structural stability.

For actions of continuous Lie groups such as  $\mathbb{R}^k$ , the above definition of local rigidity is too strict since one has to allow for small time changes induced by automorphisms of the acting group. An action  $\alpha$  of a continuous Lie group A on a manifold M is  $C^{k,r,l}$  locally rigid if any sufficiently  $C^r$  small  $C^k$  perturbation  $\tilde{\alpha}$ is  $C^l$  conjugate to a small time change of  $\alpha$ , i.e., there exists a  $C^l$  close to identity diffeomorphism  $\mathcal{H}$  and  $\rho \in \operatorname{Aut}(G)$  close to id such that instead of (1.1) one has

(1.2) 
$$\mathcal{H} \circ \alpha(\rho(g)) = \widetilde{\alpha}(g) \circ \mathcal{H}.$$

A weaker notion in the case of actions of continuous groups is *foliation rigid-ity*: one requires the diffeomorphism  $\mathcal{H}$  to map an invariant foliation  $\mathcal{F}$  (e.g., the orbit foliation  $\mathbb{O}$  of the action  $\alpha$ ) to the perturbed foliation  $\mathcal{F}'$  (e.g., the orbit foliation

 $\widetilde{\mathbb{O}}$  of the perturbed action  $\widetilde{\alpha}$ ). The foliation rigidity in the case of the orbit foliation is usually called *orbit rigidity*.  $C^{1,1,0}$  orbit rigidity is called *structural stability*.

For a broad class of hyperbolic algebraic actions of  $\mathbb{Z}^k$  and  $\mathbb{R}^k$ , k > 2,  $C^{\infty}$ local rigidity has been established in [25]; see Section 2.1 for a discussion. In this paper we prove for the first time local rigidity for a class of *partially hyperbolic* actions of  $\mathbb{Z}^k$ , namely for type (1) actions on tori.

The orbits of partially hyperbolic algebraic actions are parts of homogeneous isometric foliations. For such actions, e.g., those of type (2) differentiable rigidity should be understood in a modified sense allowing not only a linear time change in A but also an algebraic perturbation of A inside the isometric foliation; see [8], [5] for a detailed discussion.

#### 1.3. Statement of results.

1.3.1. *Rigidity*. Let  $GL(N, \mathbb{Z})$  be the group of integer  $N \times N$  matrices with determinant  $\pm 1$ . Any matrix  $A \in GL(N, \mathbb{Z})$  defines an automorphism of the torus  $\mathbb{T}^N = \mathbb{R}^N / \mathbb{Z}^N$  which we also denote by A.

An action  $\alpha: \mathbb{Z}^k \times \mathbb{T}^N \to \mathbb{T}^N$  by automorphisms is given by an embedding  $\rho_{\alpha}: \mathbb{Z}^k \to \mathrm{GL}(N, \mathbb{Z})$  so that

$$\alpha(g, x) = \rho_{\alpha}(g) x$$

for any  $g \in \mathbb{Z}^k$  and any  $x \in \mathbb{T}^N$ . We will write simply  $\alpha(g)$  for  $\rho_{\alpha}(g)$ . An action  $\alpha' : \mathbb{Z}^k \times \mathbb{T}^{N'} \to \mathbb{T}^{N'}$  is an *algebraic factor* of  $\alpha$  if there exists an epimorphism  $h: \mathbb{T}^N \to \mathbb{T}^{N'}$  such that  $h \circ \alpha = \alpha' \circ h$ .

An action  $\alpha$  by automorphisms of  $\mathbb{T}^N$  is called *irreducible* if any nontrivial algebraic factor of  $\alpha$  has finite fibers.

An action  $\alpha'$  is a *rank one factor* if it is an algebraic factor and if  $\rho_{\alpha'}(\mathbb{Z}^k)$ contains a cyclic subgroup of finite index.

An ergodic action  $\alpha$  by toral automorphisms has no nontrivial rank one factors if and only if  $\rho_{\alpha}(\mathbb{Z}^k)$  contains a subgroup isomorphic to  $\mathbb{Z}^2$  such that all nontrivial elements in this subgroup are ergodic toral automorphisms (see for example [34]). We call ergodic  $\mathbb{Z}^k$  actions by toral automorphisms *higher rank* if they have no nontrivial rank one factors.

THEOREM 1. Let  $\alpha: \mathbb{Z}^k \times \mathbb{T}^N \to \mathbb{T}^N$  be an ergodic higher rank action of  $\mathbb{Z}^k$ (k > 2) by automorphisms of the N-dimensional torus. Then there exists a constant  $l = l(\alpha, N) \in \mathbb{N}$  such that  $\alpha$  is  $C^{\infty, l, \infty}$  locally rigid.

The constant l in the above theorem depends on the dimension of the torus and the given linear action, and is precisely defined in the proof of Theorem 1 (Section 5).

We note that the result above extends to affine actions whose linear parts are actions described in Theorem 1. While any affine map of the torus without eigenvalue one in its linear part has a fixed point and hence is conjugate to its linear part, the case of commuting affine maps is more general (for examples of affine actions without fixed points see [17]). However the case of commuting affine maps reduces to the case of commuting linear maps by passing to a subgroup of finite rank which has a fixed point and observing that the centralizer of an ergodic partially hyperbolic action by toral automorphisms is discrete.

We also obtain rigidity for suspensions of actions considered in Theorem 1. (For the detailed description of the suspension construction see for example [23, §2.2] or [3, §5.4]).

THEOREM 2. Suspension of a  $\mathbb{Z}^k$  action by toral automorphisms with no rank one factors is  $C^{\infty,l,\infty}$  locally rigid (up to an automorphism of the acting group  $\mathbb{R}^k$ ) where l is the same as in Theorem 1.

Theorem 2 is deduced from Theorem 1 in Section 6. It involves a general statement (Lemma 6.1), which we present for the sake of completeness, but also relies on cocycle rigidity results which are specific.

1.3.2. *Existence*. Let us call an action of  $\mathbb{Z}^k$  by automorphisms of a torus *genuinely partially hyperbolic* if it is ergodic with respect to Lebesgue measure but no element of the action is hyperbolic (Anosov). It is easy to see that this is equivalent to simultaneous existence of

- (1) an element of the action none of whose eigenvalues is a root of unity and
- (2) an invariant linear foliation on which there is no exponential expansion/contraction for any element of the action.

As before, such an action is higher rank if and only if it contains a  $\mathbb{Z}^2$  action all of whose elements are ergodic.

The method of Theorem 1 is essential in proving rigidity of higher rank genuinely partially hyperbolic actions since for hyperbolic actions earlier methods are available [25]. The following result discusses the existence of genuinely partially hyperbolic higher rank actions.

THEOREM 3. Genuinely partially hyperbolic higher rank  $\mathbb{Z}^2$  actions exist: on any torus of even dimension  $N \ge 6$  there are irreducible examples while on any torus of odd dimension  $N \ge 9$  there are only reducible examples. There are no examples on tori of dimension  $N \le 5$  and N = 7.

*Remark* 1. In Section 9 in addition to the general construction for examples in even dimensions, we give an explicit example in dimension 6.

*Remark* 2. Using our constructions in Section 9, a similar statement as Theorem 3 about existence of genuinely partially hyperbolic higher rank actions of  $\mathbb{Z}^k$  for k > 2 can be obtained. For example, an irreducible action of such kind exists in any even dimension starting from 2k + 2.

#### 2. Old and new approaches to rigidity of abelian group actions

Present paper, although it deals with a specific problem, is a part of a broader program of studying local differentiable rigidity for hyperbolic and partially hyperbolic actions of higher rank abelian groups. Moreover, it is the first in a series of

papers by the same authors which address the partially hyperbolic situation, which is considerably more difficult that the hyperbolic one. It is natural in this context to briefly outline the developments which led to the present work, explain both successes and deficiencies of the previous methods, present the current state-of-theart, and discuss both current difficulties and prospects of the further developments.

2.1. The a priori regularity method. This method first appeared in a rudimentary fashion in [21] in the context of proving rigidity of some standard lattice actions on tori, and was introduced in its general form in [25]. It can be very briefly described as follows. One considers an action with sufficiently strong hyperbolic properties. The  $C^0$  orbit structure (or an essential part of it) of such an action is rigid (structural stability, Hirsch-Pugh-Shub theory [16]). Moreover the  $C^0$ orbit equivalence is unique transversally to the orbits. The method then consists of establishing a priori regularity of the orbit equivalence. The central idea is that *locally* in the phase space there are invariant geometric structures on invariant foliations for both the original and perturbed action. The simplest example of such a structure is a flat affine connection (local linearization); in general the structure is more complicated, but it still has a finite-dimensional Lie group of automorphisms. Such a structure appears due to the fact that some elements of the action contract the foliations in question and this is true both in the rank one and higher rank hyperbolic situations. Higher rank is crucial in showing that the conjugacy, which is *a priori* only continuous, preserves the invariant structure and hence is smooth in the direction of each foliation. This in turn relies on the fact that in certain critical directions the unperturbed action acts by isometries on each leaf of the foliation. This in particular excludes the case of actions by automorphisms of the torus and of a nil-manifold in the presence on nontrivial Jordan blocks.<sup>1</sup>

For the hyperbolic  $\mathbb{R}^k$  actions there is another step from smooth orbit equivalence to smooth conjugacy of the actions up to an automorphism of  $\mathbb{R}^k$ . This follows from the cocycle rigidity which allows to straighten out the time change [23]. Notice that cocycle rigidity also holds for some partially hyperbolic actions of higher rank abelian groups [24]. In both of those papers cocycle rigidity has been established using harmonic analysis.

This method of establishing regularity of a conjugacy (whose existence is obtained from a different kind of reasoning) also appears as an ingredient in the proofs of local rigidity for actions of certain groups other than abelian ([21], [22], [29], and [14]).

2.2. Difficulties in the partially hyperbolic case. If one tries to follow this approach for a *partially hyperbolic* algebraic action of a higher rank abelian group with semisimple linear part one starts with structural stability of the *neutral folia*tion for the unperturbed action. Then two difficulties arise. First, this foliation for perturbed action is not necessarily smooth and hence smoothness of the foliation

<sup>&</sup>lt;sup>1</sup>Added in Proof: Partial results for that case were obtained later in [11].

conjugacy along the stable and unstable directions of various elements of the action does not guarantee the global regularity. Second, even if one assumes smoothness of the neutral foliation, or, equivalently, only considers perturbations along this foliation, cocycle rigidity of the unperturbed algebraic action is not sufficient.

To overcome these difficulties we introduce two new methods: one, which is used in the present paper and has further applications, and the other, briefly outlined in Section 2.5, which is developed in [5] and further in [4].<sup>2</sup> The second method still involves the *a priori* regularity arguments as an essential component. Results of [5] and [4] are announced in [8].

*Remark* 3. Notice that for case of general lattice actions considered in [14] the partially hyperbolic case is also central. In this situation totally different methods, specific to groups with Property (T), are used.

2.3. The KAM/Harmonic analysis approach. In our new approach we do not start from a conjugacy of low regularity. Instead we construct one of high regularity by an iterative process as a fixed point of a certain nonlinear operator. We use an iterative procedure, namely an adapted version of the KAM procedure motivated by the procedure used by Moser in [31] to study small perturbations of *isometric*  $\mathbb{Z}^k$  actions generated by circle rotations. Moser first noticed that commutativity along with simultaneous Diophantine condition was enough to provide a smooth solution to certain over-determined system of equations.

Notice however, a principal difference between Moser's setting and ours: The only obstruction to solving linearized conjugacy equation in the case of commuting circle rotations is the same as for a single one: it comes from the rotation number. In our case there are infinitely many obstructions for solving the linearized equation for a single element of the action but they magically vanish for the whole action due to the "higher rank trick". Thus while Moser used his setting to show that Diophantine conditions for individual elements of the action are not necessary to contain the effects of "small denominators", we obtain differentiable conjugacy for an action in the absence of even a formal conjugacy for perturbations of its individual elements.

The solution is the limit of successive approximations obtained by approximating the nonlinear problem by its linear part, and solving (approximately) the corresponding linearized equation. This scheme arises from the classical KAM, which also set the basis for generalization of classical Implicit Function Theorem to so-called hard implicit function theorems [38], or Nash-Moser inverse function theorems [15]. (For more references see for example [10].) Those general results have been applied more recently to *isometric* "large" lattice actions. We discuss relevance of the Implicit Function Theorem approach in our situation and quote other relevant work in Section 8. We note that these general results do not apply to  $\mathbb{Z}^k$  actions generated by circle rotations studied by Moser in [31].

 $<sup>^{2}</sup>$ Added in Proof: See also two recent papers by Z. J. Wang [36], [37].

In our case the first key observation is that the linearized equation in this case is "almost" a twisted cocycle equation over the unperturbed linear action. It is possible to solve such equations due to the "higher rank trick" which lies at the root of cocycle rigidity [23], [24] and which is adapted in the current paper to the twisted case.

The second key observation is that it is possible to obtain an approximate solution to the linearized equation if the linearized equation can be approximated by a twisted cohomology equation over the unperturbed action. This reduces to approximating a certain twisted cocycle over the perturbed action by a twisted cocycle over the unperturbed action. We construct this approximation cocycle explicitly by adapting and developing a method from [18].

As a by-product of our approach we remove the restriction of semi-simplicity for the unperturbed action. Our method works when the unperturbed action, whether hyperbolic or partially hyperbolic, has nontrivial Jordan blocks. A disadvantage of our method is the requirement that the perturbation is small in  $C^{l}$  topology for a certain large l while the method of [25], when it is applicable, can deal with small perturbations in  $C^{1}$  topology.<sup>3</sup>

2.4. *Extensions of the* KAM/*harmonic analysis approach to other actions*. The method presented in this paper provides a blueprint for studying local rigidity for other algebraic abelian actions.

(I) A natural generalization of actions by automorphisms of a torus appears on infranilmanifolds when one considers projections of partially hyperbolic actions by  $\mathbb{Z}^k$  on a simply connected nilpotent group N preserving a lattice  $\Lambda$  on the compact factor-space  $N/\Lambda$ . The work on the nilpotent case is in progress.

In [25, §4.4] it is explained (although calculations are omitted) how cocycle rigidity for the nilpotent case is deduced from the cocycle rigidity in the toral case. What is needed in addition is first, tame estimates and second, the splitting. In order to do that, Fourier analysis of Section 4 is substituted by arguments using infinite-dimensional unitary representations of nilpotent groups.

(II) Some ideas from specific constructions carried out in Section 4 have been applied recently to certain totally nonhyperbolic, namely, *parabolic* abelian actions. The simplest example is the action of the upper-triangular unipotent subgroup of  $SL(2, \mathbb{R}) \times SL(2, \mathbb{R})$  on the factor by a cocompact irreducible lattice. Cocycle rigidity with tame estimates as well as tame splitting are established by D. Mieczkowski in his 2006 Ph. D. thesis and published as [30]. Such actions are not locally rigid already among homogeneous actions. In that they are similar to the rotations on the circle. However, the iteration scheme with an iteration step relying on the results

<sup>&</sup>lt;sup>3</sup>Added in Proof: M.Einsiedler and T. Fisher in [11] developed a version of the *a priori* regularity method to prove local differentiable rigidity for hyperbolic actions by automorphisms of a torus in the presence of Jordan blocks. They require an additional "narrow spectrum" assumption for proportional Lyapunov exponents.

of [30] can be carried out producing certain modified form of local rigidity restricted to parametric families of perturbations; see [9]. The number of parameters needed is the same as the codimension of unipotent subgroups within the ambient Lie group; it is equal to two in the basic example above. Due to the obstacles, just like in the case of commuting circle diffeomorphisms [31], explicit iteration is needed resembling the one carried out in Sections 5.3 and 5.4. This is in a contrast with other situations where one may use generalized implicit function theorems; see Section 8 for more discussion.

(III) Added in Proof: Very recently decisive advances have been made in extending applications of this approach to the actions of type (2) as well other algebraic partially hyperbolic actions. This confirms the power and fruitfulness of the KAM-based method. A key difference with the setting of the present paper is that one can apply a priori regularity method to obtain smooth rigidity of the neutral foliation of the action and hence reduce the problem to a perturbation along this foliation. This leads to untwisted linearized conjugacy equations and allows to use available estimates for decay of matrix coefficients. This work also contains a number of other new ingredients. (The paper by the second author and Z. J.Wang is in preparation)

2.5. *The geometric approach*. In [5], [4] we use another approach to the semisimple homogeneous case (2) which bypasses the difficulties of the KAM/ harmonic analysis method. For the sake of completeness of our account of the state of rigidity program, we present a very brief narrative outline here, referring for precise formulations and details to the papers and the published detailed announcement [8].

By the structural stability result of Hirsch-Pugh-Shub [16], the problem of obtaining a Hölder conjugacy reduces to problem of reducing a Hölder cocycle over the *perturbed action* to a constant cocycle. Obstructions to this can be described via *periodic cycle functionals* defined on closed paths along coarse Lyapunov foliations [20], [7]. For the unperturbed action such closed paths have a structure coming from the algebraic structure of the covering group, and it is proved in [7] that this structure implies vanishing of obstructions to cocycle trivialization. The main part of the geometric approach is to show that the fine structure of Lyapunov foliations for the unperturbed action is robust under small perturbations, so that it can be used to imply vanishing of obstructions to cocycle trivialization for small perturbations and produce a Hölder conjugacy with the unperturbed action. Using the *a priori* regularity method (see Section 2.1) one shows that conjugacy is smooth along the unipotent foliations for the unperturbed action. Then the usual regularity along stable directions and Hörmander theorems guarantee that the conjugacy is  $C^{\infty}$ .<sup>4</sup>

<sup>&</sup>lt;sup>4</sup>Added in Proof: This method has been successfully applied in [36], [37] to a variety of new cases of type (2) including those with nonsplit groups G. This work is based on advances in calculating appropriate families of generating relations for those groups.

2.6. *Conclusion*. Notice that for partially hyperbolic actions the geometric approach requires sufficient noncommutativity for various stable foliations for their brackets to generate the neutral directions and hence it cannot be applied in principle to partially hyperbolic actions on the torus where all invariant distributions commute. Thus the KAM and the geometric approach nicely complement each other in advancing significantly the solution of the local differentiable rigidity problem for algebraic partially hyperbolic actions.

Results of the current paper are announced in [6]. We would like to thank Elon Lindenstrauss who pointed out an inaccuracy in an early version of this paper. As a result Section 4.5 has been reworked. We also thank David Fisher by bringing to our attention the relevant work of Benveniste.

#### 3. Setting of the problem and overview of the KAM scheme

3.1. Ergodic toral automorphisms. An automorphism of the torus  $\mathbb{T}^N$  is induced by an invertible integer matrix A. We will use the same notation A both for an integer matrix and for the toral automorphism induced by it. A induces the map given by the transpose matrix  $A^t$  on the dual group  $\mathbb{Z}^N$ . If  $e_n(x) = e^{2\pi i n \cdot x}$  are the characters, then  $A^t$  acts as

$$e_{A^t n}(x) = e^{2\pi i A^t n \cdot x} = e^{2\pi i n \cdot Ax} = e_n(Ax),$$

which implies that

$$\widehat{(f \circ A)}_n = \widehat{f}_{(A^t)^{-1}n}$$

for Fourier coefficients of any  $C^{\infty}$  function f on  $\mathbb{T}^N$ . We call  $(A^t)^{-1}$  the dual map on  $\mathbb{Z}^N$ . To simplify the notation in the rest of the paper, whenever there is no confusion as to which map we refer to we will denote the dual map by the same symbol A. When we wish to clearly distinguish the two maps, we use for the dual map the notation  $A^*$ . The dual orbits of A are  $\mathbb{O}(n) = \{A^i n \mid i \in \mathbb{Z}\}$ .

The following is a simple characterization of ergodicity of toral automorphisms in terms of the corresponding dual action. Even though this characterization is a well-known fact we include also a quick proof here, for completeness.

LEMMA 3.1. (i) An automorphism of  $\mathbb{T}^N$  induced by  $N \times N$  integer matrix A is ergodic if and only if all nontrivial orbits of the dual map  $A^*$  on  $\mathbb{Z}^N$  are infinite.

(ii) An automorphism of  $\mathbb{T}^N$  induced by  $N \times N$  integer matrix A is ergodic if and only if A has no roots of unity in the spectrum.

*Proof.* (i) Suppose there exists a nonzero  $n \in \mathbb{Z}^N$  with a finite orbit, i.e.,  $A^m n = n$  for some m. Then the function  $f = \sum_{i=0}^{m-1} e_{A^i n}$  is an A invariant nonconstant function, which contradicts ergodicity.

Conversely, if A is not ergodic and if  $f = \sum_n \hat{f_n} e_n$  is an invariant function, then  $\hat{f}_{An} = \hat{f_n}$  for all  $n \in \mathbb{Z}^N$ , therefore if the orbit of n is infinite then  $\hat{f_n} = 0$ . This implies that f is constant. Thus A is ergodic. (ii) follows from (i) and from the fact that the dual map has a root of unity in the spectrum if and only if there exists a nontrivial  $n \in \mathbb{Z}^N$  with finite orbit.  $\Box$ 

3.2. *Higher rank actions by toral automorphisms*. Any action  $\alpha$  of  $\mathbb{Z}^k$  by toral automorphisms, induces a *dual action*  $\alpha^*$  on  $\mathbb{Z}^N$  given by

$$\rho_{\alpha^*}(i)n = (A_1^*)^{i_1} (A_2^*)^{i_2} \cdots (A_k^*)^{i_k} n$$

for  $i = (i_1, \ldots, i_k) \in \mathbb{Z}^k$  and  $n \in \mathbb{Z}^N$ . Then for every  $f = \sum_{n \in \mathbb{Z}^N} \hat{f}_n e_n$  we have

$$f(\rho_{\alpha}(i)x) = \sum_{n \in \mathbb{Z}^N} \hat{f}_{\alpha^i n} e_n$$

where, to simplify the notation (whenever there is no confusion), we use the notation  $\alpha^{i} n$  for the dual action  $\rho_{\alpha^{*}}(i)n$  on  $\mathbb{Z}^{N}$ .

Since we are interested in actions with no rank one factors we will assume further on that the action  $\alpha$  satisfies the following:

( $\mathfrak{G}$ ) There exist  $g_1, g_2 \in \mathbb{Z}^k$  with  $A \stackrel{\text{def}}{=} \rho_{\alpha}(g_1)$  and  $B \stackrel{\text{def}}{=} \rho_{\alpha}(g_2)$  such that any  $A^l B^k$ , for a nonzero  $(l,k) \in \mathbb{Z}^2$ , is ergodic. A and B will be referred to as ergodic generators.

The following is a simple lemma which shows that obtaining a  $C^{\infty}$  conjugacy for one ergodic generator suffices for the proof of Theorem 1.

LEMMA 3.2. Let  $\alpha$  be a  $\mathbb{Z}^k$  action by automorphisms of  $\mathbb{T}^N$  such that for some  $g \in \mathbb{Z}^k$  the automorphism  $\alpha(g)$  is ergodic. Let  $\tilde{\alpha}$  be a  $C^1$  small perturbation of  $\alpha$  such that there exists a  $C^{\infty}$  map  $H: \mathbb{T}^N \to \mathbb{T}^N$  which is  $C^1$  close to identity and satisfies

(3.1) 
$$\widetilde{\alpha}(g) \circ H = H \circ \alpha(g).$$

Then *H* conjugates the corresponding maps for all the other elements of the action; *i.e.*, for all  $h \in \mathbb{Z}^k$  we have

$$\widetilde{\alpha}(h) \circ H = H \circ \alpha(h).$$

*Proof.* Let h be any element in  $\mathbb{Z}^k$ , other than g. It follows from (3.1) and commutativity that

$$\widetilde{\alpha}(g) \circ \widetilde{H} = \widetilde{H} \circ \alpha(g),$$

where

$$\widetilde{H} = \widetilde{\alpha}(h)^{-1} \circ H \circ \alpha(h).$$

Therefore,

$$\alpha(g) \circ \tilde{H}^{-1} \circ H = \tilde{H}^{-1} \circ H \circ \alpha(g).$$

Denote the lift of  $\tilde{H}^{-1} \circ H$  to  $\mathbb{R}^N$  by id  $+\Omega$  where  $\Omega$  is a  $C^1$  small  $\mathbb{Z}^N$ -periodic  $C^{\infty}$  map, and use the same notation  $\alpha(g)$  for the lift of  $\alpha(g)$ . Then

(3.2) 
$$\alpha(g) \circ (\mathrm{id} + \Omega) = (\mathrm{id} + \Omega) \circ \alpha(g)$$

If A is diagonalizable, equation (3.2) reduces to several equations of the following type:

$$\lambda \omega = \omega \circ A,$$

where  $A = \alpha(g)$ ,  $\omega$  is  $C^{\infty}$  function, and  $\lambda$  is an eigenvalue of A. Then, passing to the dual action (while, as mentioned before, we use the same notation for the dual and for the original action), we have

$$\lambda \widehat{\omega}_n = \widehat{\omega}_{An},$$

which implies

$$\lambda^i \widehat{\omega}_n = \widehat{\omega}_{A^i n}$$

for all  $i \in \mathbb{Z}$ . Therefore

$$\sum_{i\in\mathbb{Z}}^{\infty}\lambda^{i}\widehat{\omega}_{n}=\sum_{i\in\mathbb{Z}}\widehat{\omega}_{A^{i}n}.$$

Since A is ergodic and  $\omega$  is  $C^{\infty}$  the right-hand side converges absolutely. The left-hand side can converge only if  $\hat{\omega}_n = 0$  for all  $n \neq 0$ . Since  $\lambda \neq 1$  we also have  $\hat{\omega}_0 = 0$ , i.e.,  $\omega \equiv 0$ . This implies  $\Omega \equiv 0$  and  $H \equiv \tilde{H}$ .

If there are Jordan blocks for A, the argument above is still sufficient to deduce that  $\Omega$  has to be 0. Namely, if there is, say, a 3-Jordan block, then equation (3.2) reduces to

$$\begin{pmatrix} \lambda & 1 & 0 \\ 0 & \lambda & 1 \\ 0 & 0 & \lambda \end{pmatrix} \begin{pmatrix} \omega_1 \\ \omega_2 \\ \omega_3 \end{pmatrix} \begin{pmatrix} \omega_1 \circ A \\ \omega_2 \circ A \\ \omega_3 \circ A \end{pmatrix}.$$

This implies

$$\lambda \omega_1 + \omega_2 - \omega_1 \circ A = 0,$$
  
$$\lambda \omega_2 + \omega_3 - \omega_2 \circ A = 0,$$
  
$$\lambda \omega_3 - \omega_3 \circ A = 0.$$

From the third equation above, using ergodicity of A as in the case of a simple eigenvalue, we deduce that  $\omega_3 = 0$ . Substituting into the second equation, we obtain  $\omega_2 = 0$  and finally, using this fact in the first equation above, we obtain  $\omega_1 = 0$ . One can obviously by induction obtain  $\Omega = 0$  for Jordan blocks of any dimension. Therefore  $\tilde{H} = H$  and

$$H = \alpha(h)^{-1} \circ H \circ \widetilde{\alpha}(h)$$

for an arbitrary  $h \in \mathbb{Z}^k$ .

3.3. Overview of the KAM scheme. Let  $\alpha$  be a linear action as described in Theorem 1. Let  $\tilde{\alpha}$  be its small perturbation (the topology in which the perturbation is made will become apparent from the proof). The goal is to prove the existence of a  $C^{\infty}$  map  $H: \mathbb{T}^N \to \mathbb{T}^N$  such that  $\tilde{\alpha} \circ H = H \circ \alpha$ .

One can consider the problem of finding a conjugacy as a problem of solving the following nonlinear functional equation

$$\mathcal{N}(\widetilde{\alpha}, H) \stackrel{\text{def}}{=} \widetilde{\alpha} \circ H - H \circ \alpha = 0.$$

Following the ideas of the elementary Newton method and assuming the existence of a linear structure in the neighborhood of the identity, the identity may be viewed as an approximate solution (usually called the initial guess) of the nonlinear problem. The linearization of the operator  $\mathcal{N}$  at ( $\alpha$ , id) is

$$\mathcal{N}(\tilde{\alpha}, H) = \mathcal{N}(\alpha, \mathrm{id}) + D_1 \mathcal{N}(\alpha, \mathrm{id})(\tilde{\alpha} - \alpha) + D_2 \mathcal{N}(\alpha, \mathrm{id})(\Omega) + \operatorname{Res}(\tilde{\alpha} - \alpha, \Omega) = \tilde{\alpha} - \alpha + \alpha \circ \Omega - \Omega \circ \alpha + \operatorname{Res}(\tilde{\alpha} - \alpha, \Omega),$$

where  $\Omega = H - id$ , and  $\operatorname{Res}(\tilde{\alpha} - \alpha, \Omega)$  is quadratically small with respect to  $\tilde{\alpha} - \alpha$ and  $\Omega$ . Here  $D_1 \mathcal{N}(\alpha, id)$  and  $D_2 \mathcal{N}(\alpha, id)$  denote Frechét derivatives of the map  $\mathcal{N}$ in the first and second variable, respectively, at the point  $(\alpha, id)$ . Also, since  $\alpha$  is a linear action,  $\alpha \circ \Omega$  for each generator is given simply by a matrix applied to a vector valued map. To stress this fact we will use the notation  $\alpha \Omega$  for  $\alpha \circ \Omega$ .

If one finds H so that the linear part of the equation above is zero, i.e.,

$$\alpha(H-\mathrm{id}) - (H-\mathrm{id}) \circ \alpha = -(\widetilde{\alpha} - \alpha),$$

then such *H* is a better approximate solution of the equation  $\mathcal{N}(\tilde{\alpha}, H) = 0$  than the identity is. After obtaining a better solution, the linearization procedure and solving the linearized equation may be repeated for the new perturbation leading to an even better approximation. The difficulties which arise, in particular applications of this iterative scheme, are of two kinds: one is to solve (or solve approximately) the linearized equation, and the other has to do with obtaining good estimates for the solution so that the sequence of approximate solutions produced by this scheme converges in some reasonable function space.

We now adapt this general scheme to our specific problem concerning toral automorphisms. Any map of the torus  $\mathbb{T}^N$  into itself can be lifted to the universal cover  $\mathbb{R}^N$ . For every  $g \in \mathbb{Z}^k$ , the lift of  $\alpha(g)$  is a linear map of  $\mathbb{R}^N$ , i.e., a matrix with integer entries and with determinant  $\pm 1$ , which is also denoted by  $\alpha(g)$ . The lift of  $\tilde{\alpha}(g)$  is  $\alpha(g) + R(g)$  where R(g) is an  $\mathbb{Z}^N$ -periodic map for every g, i.e., R(g)(x+m) = R(g)(x) for  $m \in \mathbb{Z}^N$ . The lift of H is id  $+\Omega$  with a  $\mathbb{Z}^N$ -periodic  $\Omega$ .

In terms of  $\Omega$  the nonlinear conjugacy problem becomes

(3.3) 
$$\alpha \Omega - \Omega \circ \alpha = -R \circ (\mathrm{id} + \Omega)$$

and the corresponding linearized equation is

(3.4) 
$$\alpha \Omega - \Omega \circ \alpha = -R.$$

If  $\Omega$  is a solution for the linearized equation (3.4) (or at least an approximate solution, i.e., it solves (3.4) with an error which is small with respect to *R*), then

one may expect that the new perturbation defined by

$$\widetilde{\alpha}^{(1)} \stackrel{\text{def}}{=} H^{-1} \circ \widetilde{\alpha} \circ H$$

where  $H = \text{Id} + \Omega$ , is much closer to  $\alpha$  than  $\tilde{\alpha}$ ; i.e., the new error

$$R^{(1)} \stackrel{\text{def}}{=} \widetilde{\alpha}^{(1)} - \alpha$$

is expected to be small with respect to the old error R.

The comparison between the two errors as usual cannot be realized in the same function space. The norm of the old error in some function space is only small compared to the norm of the new error in some larger space, thus one has some loss of regularity. The loss however is fixed; it depends only on the initial linear action and the dimension of the torus. In the proof that we present here all the maps involved will be  $C^{\infty}$ . We have loss of regularity at an individual iterative step, in the sense described above. Thus we work with the family of  $C^r$  norms  $(r \in \mathbb{N})$  on the space of  $C^{\infty}$  maps. In Section 7.1 we treat perturbations of finite regularity: we consider the space of  $C^{r_0}$  maps with fixed  $r_0 > 0$  along with a family of  $C^r$  norms for  $r \leq r_0$ . In general, if the loss of regularity is fixed and if the family of spaces used in the iterative scheme admits smoothing operators (see Section 5.1), then the interpolation inequalities hold and it is often the case (as it is here) that the iterative procedure then can be set to converge to a smooth solution of the nonlinear equation (see for example [38]).

At an individual iterative step, the new error can be expressed as

$$R^{(1)} = \tilde{\alpha}^{(1)} - \alpha = \left[\Omega \circ \tilde{\alpha}^{(1)} - \Omega \circ \alpha + R \circ (\mathrm{id} + \Omega) - R\right] + \left[R - (\alpha \Omega - \Omega \circ \alpha)\right].$$

The error term in the first bracket comes from the linearization of the problem and is easy to estimate providing  $\Omega$  is of the same order as *R*. The difficulty lies in estimating the part of the error in the second bracket, namely solving the linearized equation (3.4) approximately, with an error quadratically small with respect to *R*.

In order to understand better where commutativity of the action and its perturbation come into the picture, we look more closely into the linearized equation. Equation (3.4) actually consists of infinitely many equations corresponding to different elements  $g_i$  of the action

(3.5) 
$$\alpha(g_i)\Omega - \Omega \circ \alpha(g_i) = -R(g_i)$$

and we need a common approximate solution  $\Omega$  to all the equations above.

If such  $\Omega$  exists, then it is easy to check that due to commutativity of each pair  $\alpha(g_i)$  and  $\alpha(g_j)$ , the map  $R: G \times \mathbb{T}^N \to \mathbb{R}^N$  satisfies the following condition:

$$(3.6) \qquad \alpha(g_i)R(g_j) - R(g_j) \circ \alpha(g_i) = \alpha(g_j)R(g_i) - R(g_i) \circ \alpha(g_j).$$

At this point we need to introduce some terminology.

Definition 1. Let  $\alpha: A \times M \to M$  be an action of an abelian group A on a manifold M, let (H, \*) be a topological group and let  $\beta_0$  be a homomorphism  $\beta_0: A \to \text{Aut } H$ . Then a map  $\beta: A \times M \to H$  is called a  $\beta_0$  twisted cocycle over  $\alpha$  if

(3.7) 
$$\beta(a+b,x) = \beta(a,\alpha(b,x)) * \beta_0(a)(\beta(b,x)).$$

A  $\beta_0$  twisted cocycle  $\beta$  is a  $\beta_0$  twisted coboundary over  $\alpha$  if there exists  $P: M \rightarrow H$  such that for all  $a \in A$ 

(3.8) 
$$\beta(a, x) = \beta_0(a)(P(x)) * (P(\alpha(a, x)))^{-1}$$

If every  $C^{\infty} \beta_0$  twisted cocycle  $\beta$  over  $\alpha$  is a  $\beta_0$  twisted coboundary over  $\alpha$  via a  $C^{\infty}$  map  $P: M \to H$ , then  $\alpha$  is said to be  $C^{\infty} \beta_0$  cocycle rigid.

If  $\beta_0$  is the identity automorphism, then  $\beta_0$  is dropped from the notations above.

If  $R: \mathbb{Z}^k \times M \to \mathbb{R}^N$  satisfies (3.6), then *R* is an  $\alpha$  twisted cocycle over the action  $\alpha$  (here we use the same notation  $\alpha$  for the linear action on  $\mathbb{T}^N$  and for its lift to  $\mathbb{R}^N$ ). Then the system of equations (3.5) can be viewed as an  $\alpha$  twisted coboundary equation over the *linear* action  $\alpha$ . However, this is not quite true: *R* is an  $\alpha$  twisted cocycle over the *perturbed* action  $\tilde{\alpha}$ , not over the linear action  $\alpha$ .

LEMMA 3.3. If  $\tilde{\alpha} = \alpha + R$  is a small perturbation of  $\alpha$  then the map R is an  $\alpha$  twisted cocycle over  $\tilde{\alpha}$ .

*Proof.* As before, we use the same notation here for the linear action  $\alpha$  and for its lift to  $\mathbb{R}^N$ :

(3.9) 
$$\widetilde{\alpha}(a+b,x) = \widetilde{\alpha}(a,\widetilde{\alpha}(b,x)) = \alpha(a,\widetilde{\alpha}(b,x)) + R(a,\widetilde{\alpha}(b,x))$$
$$= \alpha(a)\widetilde{\alpha}(b,x) + R(a,\widetilde{\alpha}(b,x))$$
$$= \alpha(a)\alpha(b)x + \alpha(a)R(b,x) + R(a,\widetilde{\alpha}(b,x)).$$

On the other hand,

$$(3.10) \quad \widetilde{\alpha}(a+b,x) = \alpha(a+b,x) + R(a+b,x) = \alpha(a)\alpha(b)x + R(a+b,x).$$

This implies

(3.11) 
$$R(a+b,x) = \alpha(a)R(b,x) + R(a,\tilde{\alpha}(b,x)).$$

Thus *R* is an  $\alpha$  twisted cocycle over  $\tilde{\alpha}$ .

There are two difficulties in solving the linearized equation (3.4). First is that R is a twisted cocycle *not* over  $\alpha$  but over  $\tilde{\alpha}$  thus (3.4) is not a twisted coboundary equation over the linear action. Second is that even if (3.4) is a twisted coboundary equation, solving it requires more care than the corresponding untwisted equation. In fact, the twist produces greater loss of regularity for the solution of the twisted coboundary problem, compared to the untwisted case, as we will see in Lemma 4.2.

Lemma 3.2 shows that it is enough to produce a conjugacy for one ergodic generator. It is clear however that, in general, it is not possible to produce a  $C^{\infty}$  conjugacy for a single element of the action, since a single genuinely partially hyperbolic toral automorphism is not even structurally stable. Indeed, Lemma 4.2 in Section 4 shows that there are infinitely many obstructions to solving the linearized equation for one generator. Therefore, we consider two ergodic generators, and reduce the problem of solving the linearized equation (3.4) to solving simultaneously the following system:

(3.12) 
$$A\Omega - \Omega \circ A = -R_A,$$
$$B\Omega - \Omega \circ B = -R_B,$$

where A and B are ergodic generators:  $A := \alpha(g_1), B := \alpha(g_2)$  and

(3.13) 
$$R_A := R(g_1), \qquad R_B := R(g_2).$$

The system (3.12) splits further into several simpler systems using an appropriate basis and the fact that *A* and *B* commute. The linear problem (3.12) is reduced to several equations of the kind

(3.14) 
$$J_A \Omega - \Omega \circ A = \Theta$$
$$J_B \Omega - \Omega \circ B = \Psi,$$

where  $J_A$  is a matrix consisting of Jordan blocks of A corresponding to an eigenvalue of A,  $J_B$  is the corresponding block of B, and  $\Theta$  and  $\Psi$  are small vector valued  $\mathbb{Z}^N$ -periodic maps given by the perturbation maps  $R_A$  and  $R_B$ . In particular, if  $\lambda$ ,  $\mu$  are simple eigenvalues of A, B respectively, then we have

(3.15) 
$$\lambda \omega - \omega \circ A = \theta$$
$$\mu \omega - \omega \circ B = \psi$$

where  $\theta$  and  $\psi$  are small  $\mathbb{Z}^N$ -periodic functions. In Lemma 4.4 we show that it is possible to solve the linearized equation (3.12) and to estimate the solution with the fixed loss of regularity if the following condition is satisfied by *R*:

$$(3.16) L(R_A, R_B) \stackrel{\text{def}}{=} (R_A \circ B - BR_B) - (R_B \circ A - AR_B) = 0.$$

As mentioned above, R does not satisfy this condition. However the fact that it satisfies the twisted cocycle condition over the *perturbed* action  $\tilde{\alpha}$  implies that it almost satisfies the twisted cocycle condition (3.16); i.e., (3.16) is satisfied up to an error which is small with respect to R. More precisely, Lemma 4.7 shows that if  $\tilde{\alpha} = \alpha + R$  is a commutative action, then  $L(R_A, R_B)$  is small with respect to R.

The key step is to show that if *R* is *almost* a twisted cocycle over  $\alpha$  then *R* is *close* to an actual twisted cocycle over  $\alpha$ . In Section 4.5 we construct a projection  $\mathcal{P}R$  of *R* to the space of twisted cocycles over  $\alpha$  so that the difference  $\mathscr{C}R = \mathscr{P}R - R$  is small with respect to *R*. More precisely, in Lemma 4.6 we show

that even if  $R_A$  and  $R_B$  do not satisfy the solvability condition (3.16), it is possible to approximate both by maps  $\mathcal{P}R_A$  and  $\mathcal{P}R_B$  which satisfy the condition (3.16) with an error bounded by the size of  $L(R_A, R_B)$ .

Lemmas 4.6, 4.4 and 4.7 combined give an approximate solution to the linearized equation (3.12).

In Section 5.1 the smoothing operators are introduced to overcome the fixed loss of regularity at each iterative step and further in Section 5 the iteration process is set and is carried out, producing a  $C^{\infty}$  conjugacy which works for the  $\mathbb{Z}^2$  action generated by the two ergodic generators, thus, according to Lemma 3.2, it works for all the other elements of the  $\mathbb{Z}^k$  action  $\alpha$ .

#### 4. Approximate solution of the linearized equation

In this section we prove the existence of an approximate solution to the linearized equation (3.12). We first introduce in Section 4.1 some notation which will be used in the rest of the paper. In Section 4.2 we describe the obstructions for solving a single linear equation in (3.15), which is a twisted coboundary equation over an ergodic toral automorphism. We show, providing the obstructions vanish, that there exists a  $C^{\infty}$  solution with tame estimates. This is a version of the arguments in [35] adapted to the case of a twisted coboundary equation. An essential ingredient in the proof of Lemma 4.2, just like in [35], is an exponential growth estimate on the dual orbit of an ergodic toral automorphism which reflects arithmetic properties of ergodic toral automorphisms and was proved by Katznelson in [26].

Section 4.3 contains corresponding growth estimate for the dual orbits of a  $\mathbb{Z}^k$  higher rank ergodic action by toral automorphisms, and applications to estimating decay of Fourier coefficients for specifically defined functions which are used in Section 4.5.

In Section 4.4 we show that the obstructions to solving each equation in (3.12) vanish providing that the maps  $R_A$  and  $R_B$  satisfy (3.16). Then in Lemma 4.7 we show that the criterion (3.16) is almost satisfied, up to an error which is small with respect to R. Finally in Section 4.5 we construct a projection  $(\mathcal{P}R_A, \mathcal{P}R_B)$  such that  $(\mathcal{P}R_A, \mathcal{P}R_B)$  satisfies (3.16) and therefore one can solve tamely the system (3.12) with  $(\mathcal{P}R_A, \mathcal{P}R_B)$  instead of  $(R_A, R_B)$ . Moreover, the projection in Section 4.5 is such that the error

$$(\mathscr{C}R_A := R_A - \mathscr{P}R_A, \ \mathscr{C}R_B := R_B - \mathscr{P}R_B)$$

is small with respect to  $(R_A, R_B)$ . This produces an approximate solution to the linearized equation (3.12).

4.1. Some notation. Let A and B be the two ergodic generators for  $\alpha$ .

(1) The dual map A\* on Z<sup>N</sup> induces a decomposition of R<sup>N</sup> into expanding, neutral and contracting subspaces. We will denote the expanding subspace by V<sub>1</sub>(A), the contracting subspace by V<sub>3</sub>(A) and the neutral subspace by V<sub>2</sub>(A).
(4.1) R<sup>N</sup> = V<sub>1</sub>(A) ⊕ V<sub>2</sub>(A) ⊕ V<sub>3</sub>(A).

All three subspaces  $V_i(A)$ , i = 1, 2, 3 are A invariant and

(4.2) 
$$\|A^{i}v\| \geq C\rho^{i} \|v\|, \quad \rho > 1, \quad i \geq 0, \quad v \in V_{1}(A),$$
$$\|A^{i}v\| \geq C\rho^{-i} \|v\|, \quad \rho > 1, \quad i \leq 0, \quad v \in V_{3}(A),$$
$$\|A^{i}v\| \geq C(|i|+1)^{-N} \|v\|, \quad i \in \mathbb{Z}, \quad v \in V_{2}(A).$$

- (2) For  $n \in \mathbb{Z}^N$ ,  $|n| \stackrel{\text{def}}{=} \max\{\|\pi_1(n)\|, \|\pi_2(n)\|, \|\pi_3(n)\|\}$ , where  $\|\cdot\|$  is Euclidean norm and  $\pi_i(n)$  are projections of *n* to subspaces  $V_i(i = 1, 2, 3)$  from (4.2), that is, to the expanding, neutral, and contracting subspaces of  $\mathbb{R}^N$  for *A* (or *B*; we will use the norm which is more convenient in a particular situation; those are equivalent norms, the choice does not affect any results).
- (3) For n∈Z<sup>N</sup> we say n is mostly in i(A) for i = 1, 2, 3 and will write n → i(A), if the projection π<sub>i</sub>(n) of n to the subspace V<sub>i</sub> corresponding to A is sufficiently large:

$$|n| = \|\pi_i(n)\|.$$

The notation  $n \hookrightarrow 1, 2(A)$  will be used for *n* which is mostly in 1(A) or mostly in 2(A).

(4) Given a complex number λ and a function φ on the torus, define the twisted coboundary operators:

$$\Delta_A^\lambda \varphi \stackrel{\text{def}}{=} \lambda \varphi - \varphi \circ A, \qquad \Delta_A^\lambda \widehat{\varphi}_n \stackrel{\text{def}}{=} \lambda \widehat{\varphi}_n - \widehat{\varphi}_{An}.$$

In what follows  $\lambda$  will usually be an eigenvalue of *A*, and  $\mu$  will usually denote an eigenvalue of *B*, so we will often use the following simpler notation:

$$\Delta^{\lambda} \stackrel{\text{def}}{=} \Delta^{\lambda}_{A}, \qquad \Delta^{\mu} \stackrel{\text{def}}{=} \Delta^{\mu}_{B},$$
$$\Delta^{\lambda} \widehat{\varphi}_{n} \stackrel{\text{def}}{=} \Delta^{\lambda}_{A} \widehat{\varphi}_{n}, \qquad \Delta^{\mu} \widehat{\varphi}_{n} \stackrel{\text{def}}{=} \Delta^{\mu}_{B} \widehat{\varphi}_{n}.$$

(5) Similarly, define the following operator

$$\Delta^A \mathcal{F} \stackrel{\text{def}}{=} A \mathcal{F} - \mathcal{F} \circ A.$$

(6) For the functions  $\theta, \psi$  and the maps  $\mathcal{F}, \mathcal{G}$  define the following operators:

$$L(\theta,\psi) \stackrel{\text{def}}{=} \Delta^{\mu}_{B} \theta - \Delta^{\lambda}_{A} \psi, \qquad L(\mathcal{F},\mathcal{G}) \stackrel{\text{def}}{=} \Delta^{B} \mathcal{F} - \Delta^{A} \mathcal{G}.$$

(7) We introduce the notation for the following sums:

$$\sum_{+}^{A} \widehat{\varphi}_{n} \stackrel{\text{def}}{=} \sum_{i=0}^{\infty} \lambda^{-(i+1)} \widehat{\varphi}_{A^{i}n},$$
$$\sum_{-}^{A} \widehat{\varphi}_{n} \stackrel{\text{def}}{=} \sum_{i=-\infty}^{-1} \lambda^{-(i+1)} \widehat{\varphi}_{A^{i}n},$$
$$\sum_{-}^{A} \widehat{\varphi}_{n} \stackrel{\text{def}}{=} \sum_{i=-\infty}^{+\infty} \lambda^{-(i+1)} \widehat{\varphi}_{A^{i}n}.$$

Here *A* is used to denote the dual action  $A^*$  induced on  $\mathbb{Z}^N$  by a toral automorphism given by an integer matrix *A* (Section 3.1).  $\hat{\varphi}_n$  are Fourier coefficients of a  $C^{\infty}$  function  $\varphi$ . The corresponding notation will be used for *B* instead of *A* and  $\mu$  instead of  $\lambda$ , where  $\mu$  and  $\lambda$  are corresponding eigenvalues of *A* and *B*. We will also sometimes abbreviate the following notation as follows:

$$\sum_{l} \sum_{l=-\infty}^{def} \sum_{l=-\infty}^{-1}, \qquad \sum_{l} \sum_{l=0}^{def} \sum_{l=0}^{\infty}, \text{ and } \sum_{l=-\infty}^{l=+\infty} \sum_{l=-\infty}^{l=+\infty}.$$

(8) In what follows, C will denote any constant that depends only on the given linear action α with chosen ergodic generators and on the dimension of the torus. C<sub>x,y,z,...</sub> will denote any constant that in addition to the above depends also on parameters x, y, z,....

(9) Let 
$$f$$
 be a  $C^{\infty}$  function  $f = \sum_{n} \hat{f}_{n} e_{n}$ . Then  
(i)  $||f||_{a} \stackrel{\text{def}}{=} \sup_{n} |\hat{f}_{n}||n|^{a}$ ,  $a > 0$ .  
(ii)  $D_{j} \stackrel{\text{def}}{=} \frac{\partial}{\partial x_{j}}$ ,  $D^{k} \stackrel{\text{def}}{=} D_{1}^{k_{1}} D_{2}^{k_{2}} \cdots D_{k}^{k_{N}}$ ,  $|k| = \sum_{i=1}^{N} k_{i}$ ,  $k = (k_{1}, \dots, k_{N})$ ,  
 $||f||_{C^{r}} \stackrel{\text{def}}{=} \max_{0 \le l \le r} \sup_{\{x \in \mathbb{T}^{N}, |k| = l\}} |D^{k} fx|$ ,  $r \in \mathbb{N}_{0}$ .

(iii) The following relations hold (see, for example, Section 3.1 of [10]):

$$||f||_{r} \le C ||f||_{C^{r}}$$
 and  $||f||_{C^{r}} \le C_{r} ||f||_{r+\sigma}$ ,

where  $\sigma > N + 1$ , and  $r \in \mathbb{N}_0$ . In particular, one may take  $\sigma = N + 1 + \delta$  with a small  $\delta > 0$ .

(iv) For a map  $\mathcal{F}$  with coordinate functions  $f_i(i = 1, ..., k)$  define  $\|\mathcal{F}\|_a \stackrel{\text{def}}{=} \max_{1 \le i \le k} \|f_i\|_a$ . For two maps  $\mathcal{F}$  and  $\mathcal{G}$  define  $\|\mathcal{F}, \mathcal{G}\|_a \stackrel{\text{def}}{=} \max \{\|\mathcal{F}\|_a, \|\mathcal{G}\|_a\}$ .  $\|\mathcal{F}\|_{C^r}$  and  $\|\mathcal{F}, \mathcal{G}\|_{C^r}$  are defined similarly.

(v) *Note*. Instead of norms  $\|\cdot\|_a$  and  $\|\cdot\|_{C^r}$  one can use Sobolev norms  $\|\cdot\|_{H^s}$  for which the relation to Fourier coefficients is immediate:

$$||f||_{H^s} = \left(\sum_{n \in \mathbb{Z}} (1+|n|^2)^s |\hat{f}_n|^2\right)^{\frac{1}{2}}.$$

Then Sobolev embedding theorem would imply comparison with  $C^r$  norms with loss of  $\sigma = N/2$ . However, the explicit calculations are simpler with norms  $\|\cdot\|_a$ , and, at the expense of somewhat better estimates with Sobolev norms, we choose to use  $\|\cdot\|_a$  norms in order to keep the calculations more transparent.

(10) If  $\beta$  is a cocycle (or a twisted cocycle) over a higher rank  $\mathbb{Z}^k$ ,  $k \ge 2$  action  $\alpha$  with generators  $\alpha(g_1, \cdot)$  and  $\alpha(g_2, \cdot)$ , then we define the  $C^r$  norm of  $\beta$  by

$$\|\beta\|_{C^r} := \|\beta(g_1, \cdot), \beta(g_2, \cdot)\|_{C^r}.$$

4.2. *Twisted coboundary equation over an ergodic toral automorphism*. In what follows we make frequent use of the following lemma in which arithmetic properties of ergodic toral automorphisms play the main role. In the subsequent section we prove its generalization to higher rank actions by toral automorphisms.

LEMMA 4.1 (Katznelson, [26]). Let A be an  $N \times N$  matrix with integer coefficients. Assume that  $\mathbb{R}^N$  splits as  $\mathbb{R}^N = V \oplus V'$  with V and V' invariant under A and such that  $A|_V$  and  $A|_{V'}$  have no common eigenvalues. If  $V \cap \mathbb{Z}^N = \{0\}$ , then there exists a constant  $\gamma$  such that  $d(n, V) \geq \gamma ||n||^{-N}$  for all  $n \in \mathbb{Z}^N$  where  $|| \cdot ||$  is Euclidean norm and d is Euclidean distance.

Remark 4. This can be viewed as a version of the Liouville's theorem about rational approximation of algebraic irrationals; i.e.,  $|\alpha - \frac{m}{n}| \ge Cn^{-N}$  for any nonzero integers *m* and *n*, where  $\alpha$  is an irrational first-order root of an integer polynomial of degree *N*. The proof of this classical result inspires the proof of Lemma 4.1 in [26] which we summarize here since it gives some insight on arithmetic vs. dynamical properties of toral automorphisms.

*Proof.* Any polynomial p sufficiently close to the minimal polynomial f of A on V satisfies the condition  $p(A)n \neq 0$  for all  $n \in \mathbb{Z}^N$ ,  $n \neq 0$ , because its null space is contained in V and  $V \cap \mathbb{Z}^N = \{0\}$  by assumption. Then one can construct a polynomial  $f_Q$  with rational coefficients of that kind. The choice is made as  $|a_j - r_j/q| \leq \frac{1}{qQ}$ , where  $a_j$  are coefficients of f,  $r_j/q$  coefficients of  $f_Q$  and  $q \leq Q^k$ . Since A is an integer matrix we have  $||f_Q(A)n|| > \frac{1}{q}$  for any nonzero n. Now if  $n_V$  is the projection of n to V, then

$$f_Q(A)n = f_Q(A)(n - n_V) + (f_Q(A) - f(A))n_V.$$

This implies  $\frac{1}{q} \leq C(d(n, V) + \frac{\|n\|}{qQ})$ . Then by choosing  $Q = C \|n\|$ , where C is a constant depending on A, the estimate follows:

$$d(n, V) > \frac{1}{Cq} \ge \frac{1}{CQ^k} > C_1 ||n||^{-k} > C_1 ||n||^{-N}$$

with  $C_1$  being a positive constant depending only on A.

*Remark* 5. In particular, if A is ergodic and  $V = V_3 \oplus V_2$  from (4.1), then  $V \cap \mathbb{Z}^N = \{0\}$ . Then the above lemma implies for  $n \in \mathbb{Z}^N$ :

$$\|\pi_1(n)\| \ge \gamma \|n\|^{-N}$$

where  $\pi_1(n)$  is the projection of *n* to  $V_1$ , the expanding subspace for *A*.

Obstructions to solving a one-cohomology equation over a hyperbolic toral automorphism in  $C^{\infty}$  category are sums of Fourier coefficients of the given function along a dual orbit of the automorphism. Equivalently, obstructions are sums along the periodic orbits of the automorphism. This is the content of the Livšic theorem for hyperbolic toral automorphisms. The same characterization holds however for ergodic toral automorphisms as well, due to the estimate in Lemma 4.1. This is

due to Veech [35]. Even though the Livšic theorem generalizes to Anosov diffeomorphisms, for partially hyperbolic non-Anosov diffeomorphisms, it is a rare occurrence that such description of obstructions is possible.

We first obtain a solution to a twisted one-cohomology equation with estimates for the norm of the solution. The proof below follows closely the proof of Veech for the untwisted case [35].

LEMMA 4.2. Let  $\theta$  be a  $C^{\infty}$  function on the torus and  $\lambda \in \mathbb{C}$ ,  $\lambda \neq 1$ . Let A be an integer matrix in  $GL(N, \mathbb{Z})$  defining an ergodic automorphism of  $\mathbb{T}^N$  such that for all nonzero  $n \in \mathbb{Z}^N$ , the following sums along the dual orbits are zero, i.e.,

(4.3) 
$$\widehat{\theta}_n = \sum_{i=-\infty}^{+\infty} \lambda^{-(i+1)} \widehat{\theta}_{A^i n} = 0$$

Then the equation

(4.4) 
$$\lambda \omega - \omega \circ A = \theta$$

has a  $C^{\infty}$  solution  $\omega$ , and the following estimate:

(4.5) 
$$\|\omega\|_{a-\delta} \le \frac{C_a}{\delta^{\nu}} \|\theta\|_a$$

holds for  $\delta > 0$ ,  $\nu = aN + 1$ , and  $a > \frac{|\log |\lambda||}{\log \rho}$ . Here  $\rho > 1$  is the expansion rate for *A* from (4.2). Thus for  $r \ge 0$ 

(4.6) 
$$\|\omega\|_{C^r} \le C_r \|\theta\|_{C^{r+\sigma}},$$

where  $\sigma$  is an integer greater than  $\max\{N+1, \frac{|\lg|\lambda||}{\lg\rho}\}$ .

*Proof.* Suppose  $\omega$  is a  $C^{\infty}$  solution to (4.4) and let  $\hat{\omega}_n$  and  $\hat{\theta}_n$  denote Fourier coefficients of  $\omega$  and  $\theta$ . Then the equation  $\lambda \omega - \omega \circ A = \theta$  in the dual space has the form

$$\lambda \widehat{\omega}_n - \widehat{\omega}_{An} = \widehat{\theta}_n, \quad \forall n \in \mathbb{Z}^N.$$

For n = 0, since  $\lambda \neq 1$ , we can immediately calculate  $\hat{\omega}_0 = \frac{\hat{\theta}_0}{\lambda - 1}$ . For  $n \neq 0$  the dual equation has two solutions

$$\hat{\omega}_{n}^{(-)} = +_{(-)} \sum_{\substack{i \ge 0\\(i \le -1)}} \lambda^{-(i+1)} \hat{\theta}_{A^{i}n}, \qquad n \ne 0.$$

Each sum converges absolutely since  $\theta$  is  $C^{\infty}$  and all nonzero integer vectors have nontrivial projections to expanding and contracting subspaces for A due to the ergodicity assumption on A. By assumption  $\sum_{n=1}^{A} \hat{\theta}_n = 0$ ,  $\forall n \neq 0$ , i.e.,  $\hat{\omega}_n^+ = \hat{\omega}_n^- \stackrel{\text{def}}{=} \hat{\omega}_n$ . This gives a formal solution  $\omega = \sum_{n=1}^{A} \hat{\omega}_n^+ e_n = \sum_{n=1}^{A} \hat{\omega}_n^- e_n$ . We estimate each  $\hat{\omega}_n$  using both of its forms in order to show that  $\omega$  is  $C^{\infty}$ .

If *n* is mostly contracting, i.e., if  $n \hookrightarrow 3(A)$ , then we may use the  $\hat{\omega}_n^-$  form for the solution to obtain the following bound on the *n*-th Fourier coefficient:

$$\begin{aligned} |\widehat{\omega}_{n}| &= \left| \sum_{k \leq 0} \lambda^{-(k+1)} \widehat{\theta}_{A^{k}n} \right| \leq \sum_{k \leq 0} |\lambda|^{-(k+1)} |\widehat{\theta}_{A^{k}n}| \\ &\leq \|\theta\|_{a} \sum_{k \leq 0} |\lambda|^{-(k+1)} |A^{k}n|^{-a} \leq \|\theta\|_{a} \sum_{k \leq 0} |\lambda|^{-(k+1)} \|A^{k}\pi_{3}(n)\|^{-a} \\ &\leq \|\theta\|_{a} C^{-a} \sum_{k \leq 0} |\lambda|^{-(k+1)} \rho^{ak} \|\pi_{3}(n)\|^{-a} \leq C_{a} \|\theta\|_{a} |n|^{-a} \end{aligned}$$

where  $a > \frac{\log |\lambda|}{\log \rho}$ .

Similarly, if  $n \hookrightarrow 1(A)$ , using the form  $\hat{\omega}_n = \hat{\omega}_n^+$ , then the estimate  $|\hat{\omega}_n| \le C_a \|\theta\|_a |n|^{-a}$  holds, if  $a > \frac{\log |\lambda|^{-1}}{\log \rho}$ . If  $n \hookrightarrow 2(A)$  and  $|\lambda| \ge 1$ , using the form  $\hat{\omega}_n^+$  of the solution, then it follows

that

(4.7) 
$$\|\widehat{\omega}_{n}\| \leq \|\theta\|_{a} \sum_{k \geq 0} |\lambda|^{-(k+1)} |A^{k}n|^{-a} \\ \leq \|\theta\|_{a} C^{-a} \sum_{k \geq 0} |\lambda|^{-(k+1)} (1+k)^{N\alpha} \|\pi_{2}(n)\|^{-a}$$

However, the above sum need not converge. This is where we use again the fact that A is ergodic. Namely, according to the remark following Lemma 4.1 no integer vector can stay mostly in the neutral direction for too long. After the time which is approximately  $\ln |n|$  the expanding direction takes over. More precisely, from Lemma 4.1 it follows that  $||\pi_1(n)|| \ge \gamma |n|^{-N}$  for some  $\gamma$  and all n. Therefore

$$|A^{k}n| \ge ||A^{k}\pi_{1}(n)|| \ge C\rho^{k} ||\pi_{1}(n)|| \ge \gamma C\rho^{k} ||n|^{-N} \ge \gamma C\rho^{k-k_{0}} ||n|$$

for  $k \ge k_0$  and  $k_o = \left[\frac{(1+N)\log|n|}{\log\rho}\right] + 1$ . This fact can be used to estimate all but finitely many terms of the series in (4.7). For the rest the polynomial estimate in (4.2) for vectors in  $V_2$  holds. Hence

$$\begin{aligned} |\widehat{\omega}_{n}| &\leq \|\theta\|_{a} \sum_{k=0}^{k_{0}-1} |\lambda|^{-(k+1)} |k|^{Na} \|\pi_{2}(n)\|^{-a} \\ &+ C \|\theta\|_{a} \sum_{k=k_{0}}^{\infty} |\lambda|^{-(k+1)} \rho^{-a(k-k_{0})} |n|^{-a}. \end{aligned}$$

Thus using that  $n \hookrightarrow 2(A)$  and  $|\lambda| > 1$  we have

$$|\widehat{\omega}_n| \le C \, \|\theta\|_a |k_0|^{Na+1} |n|^{-a} + C \, \|\theta\|_a |n|^{-a}.$$

Now by choice of  $k_0$ ,  $k_0 \sim \log |n|$ . This implies the following estimate:

$$|\widehat{\omega}_n| \le C_a (\log |n|)^{Na+1} |n|^{-a} \|\theta\|_a$$

For  $|\lambda| < 1$  the same estimate follows using the second form for  $\hat{\omega}_n$ , i.e., the negative sum and the fact that  $A^{-1}$  is also an ergodic toral automorphism thus going backwards in time, the contracting direction takes over; that is, we use the Lemma 4.1 for  $A^{-1}$ . Therefore for all  $n \in \mathbb{Z}^N$  we have

$$\left|\widehat{\omega}_{n}\right|\left|n\right|^{a-\delta} \leq \frac{C_{a}}{\delta^{\nu}} \left\|\theta\right\|_{a}$$

for  $a > \frac{|\log |\lambda||}{\log \rho}$  and  $\delta > 0$ . This implies the estimate (4.5) for  $\|\omega\|_{a-\delta}$ . The estimate for  $C^r$  norms follows immediately using the norm comparison from Section 4.1. In particular if  $\theta$  is  $C^{\infty}$ , then  $\omega$  is also  $C^{\infty}$ . 

4.3. Orbit growth for the dual action. In this section the crucial estimates for the exponential growth along individual orbits of the dual action are obtained. They may be viewed as a generalization of Lemma 4.1 to higher rank actions by toral automorphisms. The implications of the growth estimates to certain estimates of the  $C^r$  norms of specifically defined functions are immediate and are formulated and proved in Lemma 4.3.

The existence of such estimates in case of  $\mathbb{Z}^d$  actions with  $d \ge 2$  relies fundamentally on the higher rank assumption.

LEMMA 4.3. Let  $\alpha$  be a  $\mathbb{Z}^d$  action by ergodic automorphisms of  $\mathbb{T}^N$ . Then there exist constants  $\tau > 0$  and C > 0 depending on the action only, such that a) For every integer vector  $n \in \mathbb{Z}^N$  and for all  $k \in \mathbb{Z}^d$ ,

$$\alpha^k n | \ge C \exp\{\tau ||k||\} |n|^{-N}.$$

b) For any  $C^{\infty}$  function  $\varphi$  on the torus, any nonzero  $n \in \mathbb{Z}^N$  and any vector  $y \in \mathbb{R}^d$  the following sums:

$$S_K(\varphi, n) = \sum_{k \in K} y^k \widehat{\varphi}_{\alpha^k n},$$

where  $y^k \stackrel{\text{def}}{=} \prod_{i=1}^d y_i^{k_i}$ , converge absolutely for any  $K \subset \mathbb{Z}^d$ .

c) Assume in addition to the assumptions in b) that for an  $n \in \mathbb{Z}^N$  and for every  $k \in K = K(n) \subset \mathbb{Z}^d$  we have  $P(||k||) |\alpha^k n| \ge |n|$  where P is a polynomial of degree N. Then

$$(4.8) |S_K(\varphi, n)| \le C_{a,y} \|\varphi\|_a |n|^{-a + \kappa_{y,\alpha}}$$

for any  $a > \kappa_{y,\alpha} \stackrel{\text{def}}{=} \frac{N+1}{\tau} \sum_{i=1}^{d} |\log |y_i||.$ d) If the assumptions of c) are satisfied for every  $n \in \mathbb{Z}^N$ , then the function

$$S(\varphi) \stackrel{\text{def}}{=} \sum_{n \in \mathbb{Z}^N} S_{K(n)}(\varphi, n) e_n$$

is a  $C^{\infty}$  function if  $\varphi$  is. Moreover, the following norm comparison holds:

(4.9) 
$$\|S(\varphi)\|_{C^r} \le C_{r,y} \|\varphi\|_{C^{r+\sigma}}$$

for r > 0 and any  $\sigma > N + 2 + [\kappa_{y,\alpha}]$ .

*Proof of* a). When d = 1, i.e., the action is given by a single ergodic toral automorphism, then the inequality a) is an immediate consequence of Lemma 4.1.

When  $d \ge 2$ , we first notice that it is enough to obtain the constant  $\tau$  and to show the exponential estimate in a) in the semisimple case, i.e., when the action is generated by matrices  $A_1, \ldots, A_d$  which are simultaneously diagonalizable over  $\mathbb{C}$ . If the action is not semisimple, only polynomial growth may occur in addition, thus the same estimate holds with slightly smaller  $\tau$  and with possibly larger *C* (for more details see [18]).

Here we give the proof in the case when the action is irreducible, which shows the main idea, but is technically simpler and we refer to [18] for the proof in the general case.

In the case when the action is irreducible, we may project a nontrivial  $n \in \mathbb{Z}^N$  to the Lyapunov directions corresponding to nonzero Lyapunov exponents of the action. Lyapunov exponents are defined as

$$\chi_i(k) = \sum_{j=1}^d k_j \ln |\lambda_{ij}|,$$

where  $k = (k_1, ..., k_d) \in \mathbb{Z}^d$ , i = 1, ..., r, and  $\lambda_{1j}, ..., \lambda_{rj}$  are the eigenvalues of  $A_j$  for j = 1, ..., d. Individual Lyapunov directions are irrational, and, due to the irreducibility assumption, each of the projections of the vector n to the Lyapunov directions is nontrivial. Thus one may apply Lemma 4.1 to each of these projections and choose  $\tau$  as the minimum of the function  $f(t) := \max_i \chi_i(t)$  for t on the unit sphere in  $\mathbb{R}^d$ . Let us assume that the minimal value  $\tau$  of the function f on the unit sphere is achieved at some point  $t^0$ .

Claim.  $\tau > 0$ .

Proof of the claim. If  $\tau \leq 0$  then for all i = 1, ..., r we have  $\chi_i(t^0) \leq 0$ . Since  $\sum_{i=1}^r \chi_i(t) = 0$  for all t, it follows that  $\chi_i(t^0) = 0$  for all i = 1, ..., rand consequently  $\tau = 0$ . Since  $t^0$  is a point on the unit circle in particular it is not zero, this implies existence of a line l in  $\mathbb{R}^d$  such that for all points on lall Lyapunov functionals take value zero. It is a result of Kronecker [27] which states that an integer matrix with all eigenvalues on the unit circle has to have all eigenvalues roots of unity. The line l then cannot contain any integer vectors  $k \in \mathbb{Z}^d$  because of the ergodicity of all the nontrivial elements  $\alpha^k$  of the action. Thus there are lattice points arbitrary close to l, in other words, there is a sequence  $k^j = (k_1^j, \ldots, k_d^j) \in \mathbb{Z}^d$  such that  $\alpha^{k_j}$  is a sequence of integer matrices whose eigenvalues all tend to 1 in absolute value as  $j \to \infty$ . At this point we may use any of the improvements of the Kronecker's result, for example [2], which implies that for a fixed dimension N of the torus, there exists a number b(N) > 1 such that any integer matrix on  $\mathbb{T}^N$  with all eigenvalues in absolute value less than b(N), has all eigenvalues roots of unity. Thus we conclude that for some j > 0 the eigenvalues of  $\alpha^{k_j}$  are roots of unity, which again contradicts the assumption on ergodicity of all elements of the action. Therefore,  $\tau$  is strictly positive.

*Note.* The constant  $\tau$  as defined above depends on the action. However by [2] it is possible to choose  $\tau$  independently of the action as  $\tau = \log(1 + \frac{1}{52N\log 6N})$ where N is the dimension of the torus.

Now that we have  $\tau > 0$ , we proceed by using the following norm:

$$\|\alpha^k n\|_{\chi} = \sum_{i=1}^r \|n_i\| \exp \chi_i(k),$$

where  $n_i$  are projections of n to the corresponding Lyapunov directions. By applying Lemma 4.1, we obtain the needed estimate

$$|\alpha^{k}n| \geq C \|\alpha^{k}n\|_{\chi} \geq C \exp\{\tau \|k\|\} \min_{i} \|n_{i}\| \geq C \exp\{\tau \|k\|\} \|n\|^{-N}.$$

*Proof of* b). The claim in b) follows from the estimate in a) and the fast decay of Fourier coefficients:

$$|S_K| \le \|\varphi\|_a \sum_{k \in K} |y|^k |\alpha^k n|^{-a} \le C_a \|\varphi\|_a |n|^{Na} \sum_{k \in K} (|y|^{\operatorname{sgn} k} e^{-a\tau})^{\|k\|}.$$

The last sum clearly converges providing  $a > \max_{i=1,\dots,d} \frac{|\log |y_i||}{\tau}$ , and for a  $C^{\infty}$ function  $\varphi$  we can choose *a* as large as needed.

*Proof of* c). Here we use the norm where  $||k|| = \max_{1 \le i \le d} |k_i|$  for k = $(k_1, \ldots, k_d) \in \mathbb{Z}^d$ . Let us choose an appropriate  $k_0 \in \mathbb{N}$  in order to split the sum  $S_{\mathcal{K}}(\varphi, n)$  into two sums: the finite one with  $||k|| < k_0$  where the polynomial estimate for  $|\alpha^k n|$  can be used, and the infinite one for  $||k|| > k_0$  where the exponential estimate obtained in a) prevails. The estimate in a) allows us to locate  $k_0$  so that it is not very large. Namely, as in Lemma 4.2 we have

$$\begin{aligned} |\alpha^{k}n| &\ge C\gamma \exp\{\tau \|k\|\} |n|^{-N} \\ &\ge C\gamma \exp\{\tau(\|k\| - k_{0})\} \exp\{\tau k_{0}\} |n|^{-N} \\ &\ge C \exp\{\tau(\|k\| - k_{0})\} |n| \end{aligned}$$

providing  $||k|| \ge k_0 > \frac{N+1}{\tau} \log |n|$ . After choosing  $k_0 = \left[\frac{N+1}{\tau} \log |n|\right] + 1$ , we split the sum  $S_K$  as follows:

$$S_{K}(\varphi, n) = S_{\|k\| < k_{0}}(\varphi, n) + S_{\|k\| \ge k_{0}}(\varphi, n)$$

To estimate the first sum we use the polynomial estimate for  $k \in K(n)$ :

$$\begin{split} |S_{||k|| < k_{0}}(\varphi, n)| &\leq C \|\varphi\|_{a} \sum_{\{k \in K: \|k\| < k_{0}\}} |y|^{k} P(\|k\|)^{a} |n|^{-a} \\ &\leq C \|\varphi\|_{a} k_{0}^{d} \prod_{i=1}^{d} \max\{|y_{i}|, |y_{i}|^{-1}\}^{k_{0}} k_{0}^{daN} |n|^{-a} \\ &\leq C_{a} \|\varphi\|_{a} k_{0}^{d(1+aN)} |n|^{-a+\kappa_{1}} \\ &\leq C_{a} \|\varphi\|_{a} (\log |n|)^{d(1+aN)} |n|^{-a+\kappa_{1}} \\ &\leq \frac{C_{a}}{\delta^{\nu}} \|\varphi\|_{a} |n|^{-a+\kappa_{1}+\delta} \end{split}$$

for any  $\delta > 0$  and any  $a > \kappa_1 = \frac{N+1}{\tau} \sum_{i=1}^{d} |\log |y_i||$ . The constant  $\nu = \nu(N, d, a)$  depends only on the dimension of the torus, rank of the action, and *a*. For the second sum we use the exponential estimate obtained in a):

$$\begin{split} |S_{||k|| \ge k_0}(\varphi, n)| &\leq C \, \|\varphi\|_a \sum_{\{k \in K: ||k|| \ge k_0\}} |y|^k |\alpha^k n|^{-a} \\ &\leq C \, \|\varphi\|_a |n|^{-a} \sum_{i=1}^d \sum_{\{k \in K: |k_i| \ge k_0\}} |y_i|^{k_i} e^{-a\tau(|k_j| - k_0)} \prod_{i \neq j=1}^{j=d} |y_j|^{k_j} e^{-a\tau|k_j|} \\ &\leq C \, \|\varphi\|_a |n|^{-a} \sum_{i=1}^d C_{a,y} \max\{|y_i|, |y_i|^{-1}\}^{k_0} \\ &\leq C_{a,y} \, \|\varphi\|_a |n|^{-a + \kappa_2} \end{split}$$

for any  $a > \kappa_2 = \frac{N+1}{\tau} \max_{1 \le i \le d} |\log |y_i||$ . By combining the estimates obtained above we have

$$|S_K(\varphi, n)| \le C_{a,y} \|\varphi\|_a |n|^{-a + \kappa_{y,a}}$$

for any  $a > \kappa_{y,\alpha} \stackrel{\text{def}}{=} \frac{N+1}{\tau} \sum_{i=1}^{d} |\log |y_i||.$ 

*Proof of* d). If for every *n* the set K(n) is chosen to satisfy the assumptions of c) then the estimate (4.8) clearly implies

$$\|S(\varphi)\|_{a-\kappa_{y,\alpha}} \le C_{a,y} \|\varphi\|_a$$

which has as a consequence the  $C^r$  estimate (4.9) with the loss of  $N + 2 + [\kappa_{y,\alpha}]$  derivatives.

4.4. Higher rank trick: solution to a twisted coboundary equation over a higher rank action by toral automorphisms. If A and B commute and there exists a solution to (3.12), then it is immediate that  $L(R_A, R_B) := \Delta^B R_A - \Delta^A R_B = 0$ . In this section we show that if every nontrivial  $A^l B^k$  is ergodic then the condition

 $L(R_A, R_B) = 0$  is not only necessary but is also sufficient for the existence of a solution to (3.12). This argument is essentially the argument used in [23] adapted to the twisted and not necessarily semisimple situation. After showing the vanishing of obstructions, the tame estimates for the common solution of (3.12) follow from Lemma 4.2.

LEMMA 4.4. If  $L(R_A, R_B) = 0$  (where  $R_A, R_B$  are  $C^{\infty}$  maps described in Section 3.3, see (3.5) and (3.13)), then the equations (3.12)

 $\Delta^A \Omega = R_A, \qquad \Delta^B \Omega = R_B$ 

have a common  $C^{\infty}$  solution satisfying

 $\|\Omega\|_{C^r} \le C_r \|R_A, R_B\|_{C^{r+\sigma}},$ 

for any r > 0 and  $\sigma > M_0 = \max\{N + 2, [m_0(A, B)]\}$ , where  $m_0(A, B)$  is a positive constant defined explicitly in (4.27) below, depending on the eigenvalues of A and B.

*Proof.* (i) *The semisimple case.* Assume that *A* and *B* are simultaneously diagonalizable. Then the equations  $L(R_A, R_B) \equiv 0$  and (3.12) split into finitely many equations of the form

(4.11) 
$$L(\theta, \psi) = \Delta^{\mu} \theta - \Delta^{\lambda} \psi \equiv 0$$

and

(4.12) 
$$\Delta^{\lambda}\omega = \theta, \qquad \Delta^{\mu}\omega = \psi,$$

where  $\theta$  and  $\psi$  are  $C^{\infty}$  functions and  $\lambda$  and  $\mu$  are corresponding eigenvalues of A and B, respectively.

The assumption  $L(\theta, \psi) \equiv 0$  implies  $\Delta^{\mu}\theta \equiv \Delta^{\lambda}\psi$ , which after passing to the dual action implies

$$\sum_{n=1}^{B} (\Delta^{\mu} \hat{\theta}_{n}) = \sum_{n=1}^{B} (\Delta^{\lambda} \hat{\psi}_{n})$$
$$\sum_{n=1}^{A} (\Delta^{\lambda} \hat{\psi}_{n}) = \sum_{n=1}^{A} (\Delta^{\mu} \hat{\theta}_{n}).$$

Consider now the first equation above. Since all the sums involved converge absolutely we have

$$\Delta^{\mu} \sum{}^{B} \widehat{\theta}_{n} = 0$$

which implies

$$\sum{}^{B}(\Delta^{\lambda}\widehat{\psi}_{n})=0.$$

Using Lemma 4.3 b), this implies that the obstruction for  $\psi$  is not only multiplied by  $\mu$  under the action of *B*, but is also multiplied by  $\lambda$  under the action of *A*, i.e.,  $\lambda \sum_{k=1}^{B} \hat{\psi}_{An} = \sum_{k=1}^{B} \hat{\psi}_{An}$ . By iterating this equation we obtain

$$\lambda^k \sum{}^B \widehat{\psi}_n = \sum{}^B \widehat{\psi}_{A^k n},$$

for every  $k \in \mathbb{Z}$ . Therefore

(4.13) 
$$\sum_{k \in \mathbb{Z}} \lambda^k \sum_{k \in \mathbb{Z}} \widehat{\psi}_n = \sum_{k \in \mathbb{Z}} \sum_{k \in \mathbb{Z}} \widehat{\psi}_{A^k n}.$$

The series in the left-hand side of (4.13) does not converge unless  $\sum_{n=1}^{B} \hat{\psi}_n = 0$  while the right-hand side of (4.13) converges absolutely by Lemma 4.3 b). Therefore  $\sum_{n=1}^{B} \hat{\psi}_n = 0$ , for all  $n \neq 0$ . Similarly  $\sum_{n=1}^{A} \hat{\theta}_n = 0$ , for all  $n \neq 0$ .

By Lemma 4.2, the formal solutions of each equation in (4.12) are  $C^{\infty}$  functions. Moreover, they coincide. Indeed, if  $\omega$  solves the second equation, i.e.,  $\Delta^{\mu}\omega = \psi$ , then

$$\Delta^{\lambda} \Delta^{\mu} \omega = \Delta^{\lambda} \psi = \Delta^{\mu} \theta.$$

Since operators  $\Delta^{\lambda}$  and  $\Delta^{\mu}$  commute this implies

$$\Delta^{\mu}(\Delta^{\lambda}\omega-\theta)=0.$$

As in Lemma 3.2 the ergodicity of *A* and *B* implies that  $\Delta^{\mu}$  and  $\Delta^{\lambda}$  are injective operators on  $C^{\infty}$ . Therefore  $\Delta^{\lambda}\omega = \theta$ , i.e.,  $\omega$  solves the first equation as well.

(ii) *The general case*. If *A* and *B* are not simultaneously diagonalizable then choose a basis in which *A* has its Jordan normal form. Since *A* and *B* commute any root space for *A* is *B* invariant. Therefore, in this basis *B* has a block diagonal form. Let  $J_A = (a_{ij})$  be an  $m \times m$  matrix which consists of blocks of *A* corresponding to the eigenvalue  $\lambda$ ; i.e., let  $a_{ii} = \lambda$  for all i = 1, ..., m and  $a_{i,i+1} = *_i \in \{0, 1\}$  for all i = 1, ..., m-1. Let  $J_B = (b_{ij})$  be the corresponding block of *B* where  $b_{ii} = \mu$  for all i = 1, ..., m ( $\mu$  is an eigenvalue of *B*) and  $b_{ij} = 0$  for all  $m \ge i > j \ge 1$ . Then because of the fact that *A* and *B* commute, by simply comparing coefficients, it is easy to obtain the following relation which the coefficients of *A* and *B* must satisfy

$$(4.14) *_i b_{ki} = *_k b_{k+1,i+1}$$

for any fixed k between 1 and m-1 and for all i = k + 1, ..., m-1.

For any such pair of blocks  $J_A$  and  $J_B$  the equations (3.12) split into equations of the form

(4.15) 
$$J_A \Omega - \Omega \circ A = \Theta$$
$$J_B \Omega - \Omega \circ B = \Psi$$

and the condition  $L(R_A, R_B) = 0$  splits as

$$(4.16) J_B \Theta - \Theta \circ B = J_A \Psi - \Psi \circ A.$$

Let the coordinate functions of  $\Theta$  and  $\Psi$  be  $\theta_i$  and  $\psi_i$  (i = 1, ..., m), respectively. Then we look for functions  $\omega_i$  (i = 1, ..., m) which solve the equations above and whose norm can be compared to the norm of  $\Theta$  and  $\Psi$ . Since  $J_A$  and  $J_B$  are upper diagonal, it is easy to obtain  $\omega_m$ . Namely, from (4.16) we have that  $\theta_m$  and  $\psi_m$  satisfy the condition  $L(\theta_m, \psi_m) = \Delta^{\mu} \theta_m - \Delta^{\lambda} \psi_m = 0$ . Therefore, using part (i), there exist  $\omega_m$  which solves simultaneously the last of m pairs of equations in (4.15), namely the equations  $\Delta^{\lambda} \omega_m = \theta_m$  and  $\Delta^{\mu} \omega_m = \psi_m$ . Moreover, the estimate

(4.17) 
$$\|\omega_m\|_{a-\delta} \le \frac{C_a}{\delta^{\nu}} \|\theta_m, \psi_m\|_a \le \frac{C_a}{\delta^{\nu}} \|\Theta, \Psi\|_a$$

follows from Lemma 4.2 for  $a \ge \max\{\frac{|\log |\lambda||}{\log \rho}, \frac{|\log |\mu||}{\log \eta}\}$  where  $\rho$  and  $\eta$  are growth rates in the hyperbolic direction corresponding to *A* and *B*, respectively. Now the (m-1)-st pair of equations in (4.15) is

(4.18) 
$$\Delta^{\lambda}\omega_{m-1} + *_{m-1}\omega_m = \theta_{m-1}$$
$$\Delta^{\mu}\omega_{m-1} + b_{m-1,m}\omega_m = \psi_{m-1},$$

while the cocycle condition for  $\theta_{m-1}$  and  $\psi_{m-1}$  from (4.16) is

(4.19) 
$$\Delta^{\mu}\theta_{m-1} + b_{m-1,m} = \Delta^{\lambda}\psi_{m-1} + *_{m-1}\psi_{m-1}.$$

By substituting  $\theta_m = \Delta^{\lambda} \omega_m$  and  $\psi_m = \Delta^{\mu} \omega_m$  into (4.19), we obtain that

$$L(\theta_{m-1} - *_{m-1}\omega_m, \psi_{m-1} - b_{m-1,m}\omega_m) = 0,$$

where the norm of both functions, on which the operator L acts, due to the estimate (4.17), can be bounded by the norm of  $\Theta$  and  $\Psi$  with a small loss. Now we may use the part (i) again to conclude that there exists some  $\omega_{m-1}$  solving the system (4.18) and such that the following estimate holds:

(4.20) 
$$\|\omega_{m-1}\|_{a-2\delta} \le \frac{C_a}{\delta^{2\nu}} \|\Theta, \Psi\|_a$$

Now we proceed by induction. Fix k between 1 and m-2 and assume that for all  $i \ge k$ , we have obtained a solution  $\omega_i$  with the appropriate estimate, i.e., for every i = k + 1, ..., m we have a  $C^{\infty}$  function  $\omega_i$  which solves the *i*-th pair of equations of (4.15):

(4.21) 
$$\Delta^{\lambda}\omega_{i} + *_{i}\omega_{i+1} = \theta_{i}$$
$$\Delta^{\mu}\omega_{i} + \sum_{l=i+1}^{m} b_{il}\omega_{l} = \psi_{i}$$

and that the following estimates hold:

(4.22) 
$$\|\omega_i\|_{a-(m-i+1)\delta} \le \frac{C_a}{\delta^{(m-i+1)\nu}} \|\Theta,\Psi\|_a$$

We wish to find  $\omega_k$  that solves the *k*-th pair of equations in (4.15):

(4.23) 
$$\Delta^{\lambda}\omega_{k} + *_{k}\omega_{k+1} = \theta_{k}$$
$$\Delta^{\mu}\omega_{k} + \sum_{i=k+1}^{m} b_{ki}\omega_{i} = \psi_{k},$$

providing that the k-th equation in (4.16) is satisfied by  $\theta_k$  and  $\psi_k$ ; i.e.,

(4.24) 
$$\Delta^{\mu}\theta_{k} + \sum_{i=k+1}^{m} b_{ki}\theta_{i} = \Delta^{\lambda}\psi_{k} + *_{k}\psi_{k+1}$$

Now we use the fact that all the subsequent pairs of equations are solved; i.e., we substitute all  $\theta_i$  for i = k + 1, ..., m and the  $\psi_{k+1}$  into (4.24) using their expression as in (4.21). This implies

$$\Delta^{\mu}\theta_{k} + \sum_{i=k+1}^{m} (b_{ki}\Delta^{\lambda}\omega_{i} + *_{i}b_{ki}\omega_{i+1})$$
$$= \Delta^{\lambda}\psi_{k} + *_{k}\Delta^{\mu}\omega_{k+1} + \sum_{i=k+1}^{m} b_{k+1,i+1}\omega_{i+1}$$

Since *A* and *B* commute, we can use the equations (4.14) for the coefficients and the linearity of operators  $\Delta^{\lambda}$  and  $\Delta^{\mu}$  to simplify the above expression to

$$\Delta^{\mu}(\theta_k - *_k \omega_{k+1}) = \Delta^{\lambda} \bigg( \psi_k - \sum_{i=k+1}^m b_{ki} \omega_i \bigg).$$

Thus the functions  $\theta_k - *_k \omega_{k+1}$  and  $\psi_k - \sum_{i=k+1}^m b_{ki} \omega_i$  satisfy the solvability condition  $L(\theta_k - *_k \omega_{k+1}, \psi_k - \sum_{i=k+1}^m b_{ki} \omega_i) = 0$ , they are  $C^{\infty}$  and therefore we may use the part (i) again to conclude that the pair of equations (4.23) has a common  $C^{\infty}$  solution  $\omega_k$ . As a consequence of assumptions (4.22) this solution satisfies the estimate

(4.25) 
$$\|\omega_k\|_{a-(m-k+1)\delta} \le \frac{C_a}{\delta^{(m-k+1)\nu}} \|\Theta,\Psi\|_a.$$

Since k is an arbitrary integer between 1 and m-1 it follows that there exists a solution  $\Omega$  to (4.15) providing that the condition (4.16) is satisfied. This can be repeated for all corresponding blocks of A and B. Since the maximal size of a Jordan block is bounded by N, we obtain the following estimate for the norm of the  $C^{\infty}$  solution  $\Omega$  of the system  $\Delta^A \Omega = R_A$ ,  $\Delta^B \Omega = R_B$ :

(4.26) 
$$\|\Omega\|_{a-N\delta} \leq \frac{C_a}{\delta^{N\nu}} \|R_A, R_B\|_a.$$

Here, *a* is any number satisfying  $a > m_0 = m_0(A, B)$ , and

(4.27) 
$$m_0(A, B) \stackrel{\text{def}}{=} \max_i \max\left\{\frac{|\log|\lambda_i||}{\log\rho}, \frac{|\log|\mu_i||}{\log\eta}\right\}$$

where the first maximum is taken over all pairs of eigenvalues  $\lambda_i$ ,  $\mu_i$  of A and B, respectively. Of course, the constant  $C_a$  has been changing throughout the procedure. It only depends on the matrices A and B and the dimension N of the torus, besides a.

As before, by fixing  $\delta$  and using the norm comparison, this implies the estimate (4.10) for  $C^r$  norm as well, with the loss of max{ $N + 2, [m_0(A, B)]$ } derivatives.

4.5. Construction of projections. As before, let  $\alpha$  denote the linear action generated by *A* and *B*. Given smooth functions  $\theta$ ,  $\psi$  and  $\varphi$ , which satisfy the equation  $L(\theta, \psi) = \varphi$ , we construct a *tame* solution for the same equation, namely we construct  $C^{\infty}$  functions  $\mathcal{E}\theta$  and  $\mathcal{E}\psi$  such that  $L(\mathcal{E}\theta, \mathcal{E}\psi) = \varphi$  and such that their norms are bounded by norms of  $\varphi$  with fixed loss of regularity. The tame estimates are obtained by using the growth estimates for the orbits of the dual action from Section 4.3.

Applying Lemma 4.5 inductively, Lemma 4.6 gives a tame solution for the equation  $L(\mathscr{E}\Theta, \mathscr{E}\Psi) = \Phi$  given that  $L(\Theta, \Psi) = \Phi$  for some  $C^{\infty}$  maps  $\Theta, \Psi, \Phi$  on the torus.

The operator *L* is linear thus in the set-up of Section 3.3, the maps  $\mathcal{P}R_A := R_A - \mathscr{C}R_A$  and  $\mathcal{P}R_B := R_B - \mathscr{C}R_B$  are  $C^{\infty}$  maps of the same order as *R*, so that the pair  $(\mathcal{P}R_A, \mathcal{P}R_B)$  is a projection of  $(R_A, R_B)$  to the space of pairs of maps which induce  $\alpha$  twisted cocycles over  $\alpha$ . In other words, the projections satisfy  $L(\mathcal{P}R_A, \mathcal{P}R_B) = 0$  and are of the same order as  $R_A$  and  $R_B$ . The appropriate estimates are contained in Lemmas 4.5 and 4.6.

*Remark.* Even though we use the same letter  $\mathcal{P}$  to denote projections of both maps  $R_A$  and  $R_B$  we note that the construction for  $\mathcal{P}R_A$  and  $\mathcal{P}R_B$  is not the same for both maps. So in this context  $\mathcal{P}$  should be viewed as a projection of the *pair*  $(R_A, R_B)$  into the space of pairs which generate  $\alpha$  twisted cocycle over the linear action  $\alpha$ .

The construction of projections gives an approximate solution to the linearized equation (3.4). Namely, since  $L(\mathcal{P}R_A, \mathcal{P}R_B) = 0$ , Lemma 4.4 implies that there exists a common smooth solution to the equations  $\Delta^A \Omega = \mathcal{P}R_A$  and  $\Delta^B \Omega = \mathcal{P}R_B$  which is of the same order as  $(R_A, R_B)$ . Then  $\Omega$  is an approximate solution of (3.12) because the error is of the order of  $(\mathcal{E}R_A, \mathcal{E}R_B)$  which is of the order of  $L(R_A, R_B)$ , and this is small with respect to  $(R_A, R_B)$  as demonstrated in Lemma 4.7. This estimate for the error is then used in Section 5 as a base for KAM iteration.

To give some intuition behind the construction of projections we recall first that from Lemma 4.4 the kernel of the operator L coincides with the space of twisted coboundaries, so the obstructions in the dual space described in Lemma 4.2 all vanish for any pair of maps in the kernel of L. Thus to construct projection to the kernel of L we simply subtract the obstructions (the part we subtract is

 $(\mathscr{E}\theta, \mathscr{E}\psi)$  and the rest  $(\mathscr{P}\theta, \mathscr{P}\psi)$  is in the kernel of *L*). The obstructions are defined on dual orbits of *B* (or *A*) and are invariant under *B* (or *A*), so we have some freedom in choosing the projection in such a way that the estimates are optimal. To construct  $(\mathscr{E}\theta, \mathscr{E}\psi)$  we concentrate the obstruction on each dual orbit for one of the generators, say B, on a single point of the dual orbit and we choose this point so that it has large expanding and large contracting components with respect to the action of *B*. Then the needed comparison that  $\mathscr{E}\psi$  projection is of the same order as  $L(\theta, \psi)$  comes from the fact that obstructions of  $\psi$  along the dual orbit of *B* are comparable to the sum of Fourier coefficients of  $L(\theta, \psi)$  along the twodimensional dual orbit of the action generated by *A* and *B*. This is where the estimates for the growth along the dual orbits of the full  $\mathbb{Z}^2$  action generated by *A* and *B* from Lemma 4.3, as well as the assumption that all nontrivial  $A^k B^l$  are ergodic, play the crucial role.

LEMMA 4.5. Let  $\theta$ ,  $\psi$ , and  $\varphi$  be  $C^{\infty}$  functions such that  $L(\theta, \psi) = \Delta^{\mu}\theta - \Delta^{\lambda}\psi = \varphi$ , then it is possible to split  $\theta$  and  $\psi$  as

$$\theta = \mathcal{P}\theta - \mathcal{E}\theta$$
$$\psi = \mathcal{P}\psi + \mathcal{E}\psi$$

so that  $L(\mathcal{P}\theta, \mathcal{P}\psi) = 0$ ,  $L(\mathcal{E}\theta, \mathcal{E}\psi) = \varphi$  and the following estimates hold:

$$(4.28) \| \mathscr{E}\theta, \mathscr{E}\psi \|_{C^r} \le C \|\varphi\|_{C^{r+o}}$$

for any r > 0 and any  $\sigma > \widetilde{M}_{\lambda,\mu}$  and

(4.29) 
$$\|\mathcal{P}\theta, \mathcal{P}\psi\|_{C^r} \le C \|\theta, \psi\|_{C^{r+\sigma}}$$

for any r > 0 and any  $\sigma > \dot{M}_{\lambda,\mu}$ . As  $\lambda$  and  $\mu$  are eigenvalues of A and B, constants  $\widetilde{M}_{\lambda,\mu}$  and  $\dot{M}_{\lambda,\mu}$  depend only on A, B and the dimension of the torus and are precisely defined below (see (4.37) and (4.44)).

*Proof.* (i) Construction of  $\mathcal{P}\theta$ ,  $\mathcal{P}\psi$ ,  $\mathcal{E}\theta$  and  $\mathcal{E}\psi$ . Let us call  $n \in \mathbb{Z}^2$  minimal and denote it by  $n_{\min}$  if n is the lowest point on its B orbit in the sense that  $n \hookrightarrow 3(B)$  and  $Bn \hookrightarrow 1, 2(B)$  (for the definition of " $\hookrightarrow$ ", see Section 4.1). There is one such minimal point on each nontrivial dual B orbit, we choose one on each dual B orbit and denote it by  $n_{\min}$ .

Now let  $\mathscr{E}\psi \stackrel{\text{def}}{=} \sum \widehat{\mathscr{E}\psi_n} e_n$  where

(4.30) 
$$\widehat{\mathscr{E}\psi_n} \stackrel{\text{def}}{=} \begin{cases} \mu \sum^B \widehat{\psi}_n, & n = n_{\min}, \\ 0, & \text{otherwise} \end{cases}$$

for  $n \neq 0$  and  $\widehat{\mathscr{C}\psi_0} \stackrel{\text{def}}{=} 0$ . Let  $\omega \stackrel{\text{def}}{=} \sum \widehat{\omega}_n e_n$  where

$$\widehat{\omega}_n \stackrel{\text{def}}{=} \begin{cases} \sum_{+}^{B} \widehat{\psi}_n, & n \hookrightarrow 1, 2(B) \\ -\sum_{-}^{B} \widehat{\psi}_n, & n \hookrightarrow 3(B) \end{cases}$$

for  $n \in \mathbb{Z}^N \setminus \{0\}$  and  $\widehat{\omega}_0 = (\mu - 1)^{-1} \widehat{\psi}_0$ . Let (4.31)  $\mathscr{P}\psi \stackrel{\text{def}}{=} \Delta^{\mu}\omega = \mu\omega - \omega \circ B.$ 

Then it is easy to check that

 $\psi = \mathcal{P}\psi + \mathcal{E}\psi.$ 

In (ii) and (iii) below we will show that both  $\mathcal{P}\psi$  and  $\mathcal{E}\psi$  are smooth functions such that  $\mathcal{P}\psi$  is of the order of  $\psi$ , and  $\mathcal{E}\psi$  is the order of  $\varphi$ .

Let us define  $\mathcal{P}\theta$  as

(4.32) 
$$\mathscr{P}\theta \stackrel{\text{def}}{=} \Delta^{\lambda}\omega$$

Then  $L(\mathcal{P}\theta, \mathcal{P}\psi) = 0$  since operators  $\Delta^{\lambda}$  and  $\Delta^{\mu}$  commute due to the commutativity of the generators A and B. Therefore, by defining  $\mathcal{E}\theta$  as

we obtain  $L(\mathcal{E}\theta, \mathcal{E}\psi) = \varphi$ ; i.e.,

(4.34) 
$$\Delta^{\mu} \mathscr{E} \theta = \Delta^{\lambda} \mathscr{E} \psi + \varphi.$$

Since operators  $\Delta^{\mu}$  and  $\Delta^{\lambda}$  are bounded, if  $\mathscr{E}\psi$  is proved to be smooth with norm comparable to some norm of  $\varphi$ , then by Lemma 4.2 the same holds true for  $\mathscr{E}\theta$  as a solution of equation (4.34). (The operator  $\Delta^{\mu}$  is injective on  $C^{\infty}$  whenever  $\mu \neq 1$ . This fact is contained in the proof of the Lemma 3.2 and is a consequence of the ergodicity of *B*.)

(ii) *Estimates for*  $\mathcal{P}\theta$  and  $\mathcal{P}\psi$ . The following estimate follows from the definition of  $\omega$  in (i) and is obtained as in the proof of Lemma 4.2:

(4.35) 
$$\|\omega\|_{a-\delta-\kappa_{\mu,B}} \le \frac{C_a}{\delta^{\nu}} \|\psi\|_a$$

for any  $a > \kappa_{\mu,B}$  and any  $\delta > 0$ , where

(4.36) 
$$\kappa_{\mu,B} \stackrel{\text{def}}{=} \frac{N+1}{\log \eta} |\log |\mu||.$$

The extra loss of  $\kappa_{\mu,B}$  appears here because the obstructions  $\sum^{B} \hat{\psi}$  do not vanish and  $|\mu|$  may be different than 1, just as in the proof of Lemma 4.3, parts b) and c). Here,  $\eta > 1$  is the constant coming from the exponential growth in the hyperbolic direction for *B* (as  $\rho$  is for *A* in (4.2)). Since operators  $\Delta^{\mu}$  and  $\Delta^{\lambda}$  are bounded in any  $\|\cdot\|_{a}$  norm, this also implies the following estimate for  $\mathcal{P}\psi$  and  $\mathcal{P}\theta$ :

$$\|\mathscr{P}\psi,\mathscr{P}\theta\|_{a-\delta-\kappa_{\mu,B}} \leq \frac{C_a}{\delta^{\nu}} \|\psi\|_a,$$

which, in particular, implies the corresponding estimate (4.29) for  $C^r$  norms for  $\mathcal{P}\psi$  and  $\mathcal{P}\theta$  with the loss of

(4.37) 
$$\dot{M}_{\lambda,\mu} \stackrel{\text{def}}{=} N + 2 + [\kappa_{\mu,B}]$$

derivatives.

(iii) *Estimates for*  $\mathscr{E}\psi$  and  $\mathscr{E}\theta$ . To estimate  $\mathscr{E}\psi$  we need to bound  $\sum_{n=1}^{B} \hat{\psi}_{n}$  in case  $n \hookrightarrow 3(B)$  and  $Bn \hookrightarrow 1, 2(B)$  with respect to  $\varphi$ . Since  $\Delta^{\mu}\theta = \Delta^{\lambda}\psi + \varphi$ , the obstructions for  $\Delta^{\lambda}\psi + \varphi$  with respect to *B* vanish; therefore

$$\Delta^{\lambda} \sum{}^{B} \widehat{\psi}_{n} = -\sum{}^{B} \widehat{\varphi}_{n}.$$

Iterating this equation with respect to A we obtain

$$\sum_{l \to \infty} {}^{B} \hat{\psi}_{n} + \lambda^{-l} \lim_{l \to \infty} \sum_{l \to \infty} {}^{B} \lambda^{-l} \hat{\psi}_{A^{l}n} = -\sum_{i=0}^{l} \sum_{l=0} {}^{B} \hat{\varphi}_{A^{i}n}.$$

From Lemma 4.3 b) the limit above is 0. By iterating backwards and applying the same reasoning, we obtain

(4.38) 
$$\sum_{n=1}^{B} \hat{\psi}_{n} = \sum_{n=1}^{A} \sum_{n=1}^{B} \hat{\varphi}_{n} = -\sum_{n=1}^{A} \sum_{n=1}^{B} \hat{\varphi}_{n}$$

In the notation of Lemma 4.3 c), (4.38) implies that for  $n \in \mathbb{Z}^N$  which is minimal on its *B* orbit, we have

$$\widehat{\mathscr{E}\psi_n} = S_{H^+}(\varphi, n) = -S_{H^-}(\varphi, n),$$

where  $H^+$  is the set of lattice points (l, k) in  $\mathbb{Z}^2$  with positive l and  $H^-$  the set of points with negative l. Then according to Lemma 4.3 d), the needed estimate for  $\mathscr{C}\psi$  with respect to  $\varphi$  follows if in at least one of the half-spaces  $H^-$  and  $H^+$  the dual action satisfies some polynomial lower bound for every  $n = n_{\min}$ .

In case  $Bn \hookrightarrow 2(B)$  for all l and all k we obviously have

(4.39) 
$$|A^{l}B^{k}n| \ge C|l|^{-N}|k|^{-N}|n|;$$

thus the polynomial estimate needed for the application of part c) of Lemma 4.3 is satisfied both in  $H^+$  and  $H^-$  for such *n*.

However in the other case, i.e., when  $Bn \hookrightarrow 1(B)$ , the same estimate holds either in  $H^+$  or in  $H^-$ . This follows from the fact that in this case  $(n \hookrightarrow 3(B)$  and  $Bn \hookrightarrow 1(B))$ , *n* is substantially large both in the expanding and in the contracting direction for *B*.

To see this we let  $n_{i_1}$  and  $n_{i_3}$  be (large) projections of *n* to some expanding and contracting Lyapunov subspaces  $V_{i_1}$  and  $V_{i_2}$  for *B* with Lyapunov exponents  $\chi_{i_1}$  and  $\chi_{i_2}$ , respectively; i.e., let

$$||n_{i_1}|| \ge C|n|$$
 and  $||n_{i_3}|| \ge C|n|$ ,

where C is some fixed positive number. Then (assuming for the moment that  $\alpha$  is semisimple) this implies

(4.40) 
$$|A^{l}B^{k}n| \geq C \sum_{i=1}^{r} \exp \chi_{i}(l,k) ||n_{i}|| \\ \geq C(\exp \chi_{i_{1}}(l,k) + \exp \chi_{i_{3}}(l,k))|n|.$$

Now we notice that the union *H* of half-spaces  $\{(l,k) : \chi_{i_1}(l,k) \ge 0\}$  and  $\{(l,k) : \chi_{i_3}(l,k) \ge 0\}$  covers either  $H^+$  or  $H^-$ . Namely, for any  $k \in \mathbb{Z}$ , (l,k) is in *H* if  $l\left(\frac{\log|\lambda_{i_1}|}{\log|\mu_{i_1}|} - \frac{\log|\lambda_{i_3}|}{\log|\mu_{i_3}|}\right) \ge 0$  and this is true for  $l \ge 0$  or for  $l \le 0$  depending on the  $\operatorname{sgn}\left(\frac{\log|\lambda_{i_1}|}{\log|\mu_{i_1}|} - \frac{\log|\lambda_{i_3}|}{\log|\mu_{i_3}|}\right)$ . Here  $\lambda_{i_3}, \lambda_{i_1}$  and  $\mu_{i_3}, \mu_{i_1}$  are corresponding eigenvalues of *A* and *B*, respectively. Therefore, from (4.40) we obtain

$$|A^l B^k n| \ge C |n|$$

in  $H^+$  or in  $H^-$  if  $\alpha$  is semisimple. If  $\alpha$  is not semisimple, then it decomposes a product of its semisimple and its unipotent part. For the semisimple part we use the estimate above and in the unipotent part only a polynomial growth may occur. This implies that (4.39) holds in  $H^+$  or in  $H^-$  for a general (not necessarily semisimple)  $\alpha$ .

Now choose the half-space in which the estimate (4.39) holds, that is choose one of the sums  $S_{H^+}(\varphi, n)$  or  $S_{H^-}(\varphi, n)$ . Then the assumptions of d) in Lemma 4.3 are satisfied for one of the sums above  $S_{H^+}$  or  $S_{H^-}$  and therefore the estimate for  $\mathscr{C}\psi$  follows:

(4.41) 
$$\|\mathscr{E}\psi\|_{a-\delta-\kappa_{(\lambda,\mu),\alpha}} \leq \frac{C_a}{\delta^{\nu}} \|\varphi\|_a$$

for any  $a > \kappa_{(\lambda,\mu),\alpha}$  and any  $\delta > 0$ , where

(4.42) 
$$\kappa_{(\lambda,\mu),\alpha} \stackrel{\text{def}}{=} \frac{N+1}{\tau} (|\log|\mu|| + |\log|\lambda||).$$

Here,  $\tau = \tau(A, B) > 0$  is the constant chosen as in the Lemma 4.3 a).

As we mentioned in part (i), by construction we have  $\Delta^{\mu} \mathcal{E} \theta = \Delta^{\lambda} \mathcal{E} \psi + \varphi$ . This by using Lemma 4.2 implies the following estimate for  $\mathcal{E} \theta$  with respect to  $\varphi$ :

(4.43) 
$$\|\mathscr{E}\theta\|_{a-\delta-\kappa_{(\lambda,\mu),\alpha}} \leq \frac{C_a}{\delta^{\nu}} \|\varphi\|_{a-\delta-\kappa_{(\lambda,\mu),\alpha}} \leq \frac{C_a}{\delta^{\nu}} \|\varphi\|_{a-\delta-\kappa_{(\lambda,\mu),\alpha}}$$

for any  $a > \kappa_{(\lambda,\mu),\alpha}$  and any  $\delta > 0$ . This implies the  $C^r$  estimate (4.28) for  $\mathscr{C}\psi$ and  $\mathscr{C}\theta$  with the loss  $\sigma > \widetilde{M}_{\lambda,\mu}$ , where

(4.44) 
$$\widetilde{M}_{\lambda,\mu} \stackrel{\text{def}}{=} N + 2 + [\kappa_{(\lambda,\mu),\alpha}]$$

where  $\kappa_{(\lambda,\mu),\alpha}$  is defined in (4.42).

LEMMA 4.6. For two  $C^{\infty}$  maps  $R_A$  and  $R_B$  with  $L(R_A, R_B) = \Phi$ , there exists a splitting

$$R_A = \mathcal{P}R_A + \mathcal{E}R_A$$
$$R_B = \mathcal{P}R_B + \mathcal{E}R_B$$

such that

$$L(\mathscr{P}R_A, \mathscr{P}R_B) = 0, \qquad L(\mathscr{C}R_A, \mathscr{C}R_B) = \Phi$$

$$(4.45) \qquad \qquad \|\mathscr{P}R_A, \mathscr{P}R_B\|_{C^r} \le C_r \|R_A, R_B\|_{C^{r+\sigma}}$$

(4.46)  $\|\mathscr{E}R_A, \mathscr{E}R_B\|_{C^r} \le C_r \|\Phi\|_{C^{r+\sigma}}$ 

for any r > 0 and  $\sigma > M = M(A, B, N)$ , where constant M depends only on the dimension of the torus and the linear action and is defined below (see (4.60)).

*Proof.* If A and B are semisimple, then the statement follows directly from Lemma 4.5 as the condition  $L(R_A, R_B) = \Phi$  splits into finitely many equations of the type

$$\Delta^{\mu}\theta - \Delta^{\lambda}\psi = \varphi,$$

where  $\theta$ ,  $\psi$ , and  $\varphi$  are  $C^{\infty}$  functions.

Now assume that A and B are not simultaneously diagonalizable and choose a basis in which A is in its Jordan normal form with some nontrivial Jordan blocks. Then in the same basis B has block diagonal form and as in Lemma 4.4 we can take  $m \times m$  blocks  $J_A$  and  $J_B$  corresponding to eigenvalues  $\lambda$  and  $\mu$  of A and B, respectively, and split  $L(R_A, R_B) = \Phi$  into equations

(4.47) 
$$J_B \Theta - \Theta \circ B - J_A \Psi + \Psi \circ A = \Phi,$$

where, as in Lemma 4.4, we take  $J_A = (a_{ij})$  to be an  $m \times m$  matrix which consists of blocks of *A* corresponding to the eigenvalue  $\lambda$ , i.e.,  $a_{ii} = \lambda$  for all i = 1, ..., m and  $a_{i,i+1} = *_i \in \{0, 1\}$  for all i = 1, ..., m-1 and  $J_B = (b_{ij})$  to be the corresponding block of *B* where  $b_{ii} = \mu$  for all i = 1, ..., m ( $\mu$  is an eigenvalue of *B*) and  $b_{ij} = 0$  for all  $m \ge i > j \ge 1$ . Equation (4.47) splits into *m* equations. For every k = 1, ..., m we have the following equation which we call  $(EQ)_k$ :

(4.48) 
$$\left(\Delta^{\mu}\theta_{k} + \sum_{i=k+1}^{m} b_{ki}\theta_{i}\right) - (\Delta^{\lambda}\psi_{k} + *_{k}\psi_{k+1}) = \varphi_{k}$$

where  $\theta_i$ ,  $\psi_i$ , and  $\varphi_i$  are coordinate functions of  $\Theta$ ,  $\Psi$ , and  $\Phi$ , respectively, in the basis in which A is in its Jordan normal form. In the special case when  $\varphi_k = 0$ , then we denote equation (4.48) by  $(EQ)_k^0$ . Clearly, for k = m the equation  $(EQ)_m$  is simply

$$\Delta^{\mu}\theta_m - \Delta^{\lambda}\psi_m = \varphi_m,$$

which by Lemma 4.5 implies the existence of the splitting

(4.49) 
$$\theta_m = \mathcal{P}\theta_m + \mathcal{E}\theta_m = \Delta^{\lambda}\omega_m + \mathcal{E}\theta_m$$
$$\psi_m = \mathcal{P}\psi_m + \mathcal{E}\psi_m = \Delta^{\mu}\omega_m + \mathcal{E}\psi_m$$

where  $\omega_m$ ,  $\mathscr{E}\theta_m$ ,  $\mathscr{E}\psi_m$ ,  $\mathscr{P}\theta_m = \Delta^{\lambda}\omega_m$ , and  $\mathscr{P}\psi_m = \Delta^{\mu}\omega_m$  are  $C^{\infty}$  functions satisfying the estimates:

DANIJELA DAMJANOVIĆ and ANATOLE KATOK

$$(4.50) \|\mathscr{E}\theta_m, \mathscr{E}\psi_m\|_{a-(\delta+\kappa)} \le \frac{C_a}{\delta^{\nu}} \|\Phi\|_a, \\ \|\omega_m\|_{a-(\delta+\kappa)} \le \frac{C_a}{\delta^{\nu}} \|\Theta, \Psi\|_a, \\ \|\mathscr{P}\theta_m, \mathscr{P}\psi_m\|_{a-(\delta+\kappa)} \le \frac{C_a}{\delta^{\nu}} \|\Theta, \Psi\|_a.$$

where we let  $\kappa \stackrel{\text{def}}{=} \max\{\kappa_{(\lambda,\mu),\alpha}, \kappa_{\mu,B}\}$  (see the proof of Lemma 4.5, (4.42), (4.36)).

Now we proceed by induction. Fix k between 1 and m-1 and assume that for all i = k + 1, ..., m we already have the splitting

(4.51) 
$$\theta_i = \mathcal{P}\theta_i + \mathcal{E}\theta_i = \Delta^{\lambda}\omega_i + \mathcal{E}\theta_i,$$
$$\psi_i = \mathcal{P}\psi_i + \mathcal{E}\psi_i = \Delta^{\mu}\omega_i + \mathcal{E}\psi_i$$

where  $\omega_i$ ,  $\mathcal{P}\theta_i = \Delta^{\lambda}\omega_i$ ,  $\mathcal{P}\psi_i = \Delta^{\mu}\omega_i$ ,  $\mathcal{E}\theta_i$  and  $\mathcal{E}\psi_i$  are  $C^{\infty}$  functions satisfying the following estimates:

$$(4.52) \qquad \|\mathscr{E}\theta_{i}, \mathscr{E}\psi_{i}\|_{a-(m-i+1)(\delta+\kappa)} \leq \frac{C_{a}}{\delta^{(m-i+1)\nu}} \|\Phi\|_{a}, \\ \|\omega_{i}\|_{a-(m-i+1)(\delta+\kappa)} \leq \frac{C_{a}}{\delta^{(m-i+1)\nu}} \|\Theta,\Psi\|_{a}, \\ \|\mathscr{P}\theta_{i}, \mathscr{P}\psi_{i}\|_{a-(m-i+1)(\delta+\kappa)} \leq \frac{C_{a}}{\delta^{(m-i+1)\nu}} \|\Theta,\Psi\|_{a},$$

and such that  $\mathcal{P}\theta_i$  and  $\mathcal{P}\psi_i$  satisfy the equation  $(EQ)_i^0$  (4.48).

By substituting from (4.51) the expressions for  $\theta_i$  and  $\psi_i$  for all i = k + 1, ..., m into (4.48), we obtain (4.53)

$$\Delta^{\mu}(\theta_k - *_k \omega_{k+1}) - \Delta^{\lambda}(\psi_k - \sum_{i=k+1}^m b_{ki}\omega_i) = \varphi_k - *_k \mathscr{E}\psi_{k+1} - \sum_{i=k+1}^m b_{ki}\mathscr{E}\theta_i.$$

Then using Lemma 4.5 again we can obtain  $\omega_k$ ,  $\mathcal{E}\theta_k$  and  $\mathcal{E}\psi_k$  such that

(4.54) 
$$\theta_k - *_k \omega_{k+1} = \Delta^\lambda \omega_k + \mathscr{E} \theta_k$$
$$\psi_k - \sum_{i=k+1}^m b_{ki} \omega_i = \Delta^\mu \omega_k + \mathscr{E} \psi_k$$

with estimates for  $\mathscr{E}\theta_k$  and  $\mathscr{E}\psi_k$  following from (4.52) and (4.53):

$$(4.55) \qquad \|\mathscr{E}\theta_{k}, \mathscr{E}\psi_{k}\|_{a-(m-k+1)(\delta+\kappa)} \leq \frac{C_{a}}{\delta^{(m-k+1)\nu}} \|\Phi\|_{a}, \\ \|\omega_{k}\|_{a-(m-k+1)(\delta+\kappa)} \leq \frac{C_{a}}{\delta^{(m-k+1)\nu}} \|\Theta,\Psi\|_{a}, \\ \|\mathscr{P}\theta_{k}, \mathscr{P}\psi_{k}\|_{a-(m-k+1)(\delta+\kappa)} \leq \frac{C_{a}}{\delta^{(m-k+1)\nu}} \|\Theta,\Psi\|_{a}.$$

where we define  $\mathcal{P}\theta_k$  and  $\mathcal{P}\psi_k$  as

(4.56) 
$$\mathcal{P}\theta_k = \Delta^\lambda \omega_k + *_k \omega_{k+1},$$
$$\mathcal{P}\theta_k = \Delta^\lambda \omega_k + \sum_{i=k+1}^m b_{ki} \omega_i.$$

Now checking that  $\mathcal{P}\theta_k$  and  $\mathcal{P}\psi_k$  are "good", i.e., that they satisfy the equation  $(EQ)_k^0$ , is easily done just by substitution. At this point however one has to use the coefficients relations (4.14) which were derived in Lemma 4.4 from the fact that A and B commute.

Since the maximal size of a Jordan block of *A* is less than *N*, we can estimate  $\mathscr{E}\Theta$  and  $\mathscr{E}\Psi$  in the  $\|\cdot\|_{a-N(\delta+\kappa)}$  norm

(4.57) 
$$\|\mathscr{E}\Theta, \mathscr{E}\Psi\|_{a-N(\delta+\kappa)} \le \frac{C_a}{\delta^{N_{\nu}}} \|\Phi\|_a$$

and similarly for the maps  $\mathcal{P}\theta$  and  $\mathcal{P}\psi$ :

(4.58) 
$$\|\mathscr{P}\Theta, \mathscr{P}\Psi\|_{a-N(\delta+\kappa)} \le \frac{C_a}{\delta^{N\nu}} \|\Theta, \Psi\|_a.$$

If repeated for all Jordan blocks, this produces the required splitting

$$(4.59) R_A = \mathcal{P}R_A + \mathcal{C}R_A R_B = \mathcal{P}R_B + \mathcal{C}R_B,$$

which satisfies the conditions and the estimates in the statement. Since we use repeatedly Lemma 4.5 the  $C^r$  estimates (4.45) and (4.46) hold for  $\sigma > M = M(A, B, N)$ , where

(4.60) 
$$M \stackrel{\text{def}}{=} N + 2 + N [\max\{\max_{i} \kappa_{\mu_{i},B}, \max_{i} \kappa_{(\lambda_{i},\mu_{i}),\alpha}\}],$$

where the first maximum in *i* is taken over all eigenvalues  $\mu_i$  of *B* (here  $\kappa_{\mu_i,B}$  are numbers defined in (4.36)) and the second maximum in *i* is taken over all pairs of corresponding eigenvalues  $\lambda_i$  and  $\mu_i$  of *A* and *B* (here  $\kappa_{(\lambda_i,\mu_i),\alpha}$  are the numbers defined in (4.42)).

The following lemma shows that  $\Phi = L(R_A, R_B)$  cannot be large if  $\alpha + R$  is a commutative action. It is in fact quadratically small with respect to R.

LEMMA 4.7. If  $\tilde{\alpha} = \alpha + R$  is a commutative  $C^{\infty}$  action of an abelian group *A* by toral automorphisms, with  $\alpha$  linear, then for  $r \geq 0$ 

(4.61) 
$$\|L(R_A, R_B)\|_{C^r} \le C_r \|R_A, R_B\|_{C^r} \|R_A, R_B\|_{C^{r+1}}$$

where  $R_A = R(g_1), R_B = R(g_2)$  and  $g_1, g_2 \in A$ .

Proof.

$$\widetilde{\alpha}_A \circ \widetilde{\alpha}_B = \widetilde{\alpha}_B \circ \widetilde{\alpha}_A$$
$$(A + R_A) \circ (B + R_B) = (B + R_B) \circ (A + R_A)$$
$$R_A \circ (B + R_B) - BR_A = R_B \circ (A + R_A) - AR_B.$$

Therefore,

$$L(R_A, R_B) = -R_A \circ B + BR_A + R_B \circ A - AR_B$$
  
= - (R\_B \circ (A + R\_A) - R\_B \circ A) + (R\_A(B + R\_B) - R\_A \circ B).

Then from the Taylor's formula with integral remainder

$$R_B \circ (A + R_A) - R_B \circ A = \int_0^1 DR_B (A + tR_A) R_A dt$$

just as in [31], it follows that

$$||R_B \circ (A + R_A) - R_B \circ A||_{C^0} \le C_r ||R_B||_{C^1} ||R_A||_{C^0}$$

A similar estimate holds for  $R_A(B + R_B) - R_A \circ B$ .

The estimate (4.61) for  $C^r$  norms, even if a bit less obvious, follows similarly (see for example [28, Appendix II]):

$$\begin{aligned} \|L(R_A, R_B)\|_{C^r} &\leq C_r(\|R_B\|_{C^{r+1}}\|R_A\|_{C^r} + \|R_A\|_{C^{r+1}}\|R_B\|_{C^r}) \\ &\leq C_r\|R_A, R_B\|_{C^r}\|R_A, R_B\|_{C^{r+1}}. \end{aligned}$$

# 5. Proof of Theorem 1

Assuming  $\tilde{\alpha}$  is a  $C^{\infty}$  perturbation of  $\alpha$ , and that the difference  $R = \tilde{\alpha} - \alpha$  is small in some  $C^{l}$  norm (where *l* is fixed and will be determined in the proof; see (5.12)) we show that  $\tilde{\alpha}$  is smoothly conjugate to  $\alpha$ . The conjugacy is produced for the two ergodic generators and by Lemma 3.2 it works for all elements of the action. This proof is similar to the iterative proof in [31]. We refer to Section 8 for a short discussion on possible variations of the proof below and possible applications of generalized implicit function theorems in this set-up.

5.1. *Smoothing*. Following the scheme described in Section 3.3 at each step of the iterative procedure we solve the linearized equation (3.12)

$$A\Omega - \Omega \circ A = -R_A, \qquad B\Omega - \Omega \circ A = -R_B$$

approximately. By results of the Section 4.5 we have that the linearized equation (3.12) has an approximate solution  $\Omega$  which is  $C^{\infty}$  although we can only compare its norm in  $C^r$  to the norm of R in  $C^{r+\sigma}$  where  $\sigma$  is large but fixed. The error is  $\mathscr{C}R$  and is by construction in Section 4.5, comparable to  $L(R_A, R_B)$ . Thus it is small with respect to R by Lemma 4.7, but comparison again comes with fixed loss of derivatives. The loss of derivatives might be a large number (depending of the hyperbolicity properties of the linear action), but it only depends on the dimension of the torus and the *unperturbed* linear action. To overcome this fixed loss of derivatives at each step of the iteration process, it is standard (see for example [38]) to introduce a family of smoothing operators  $\{S_J, J \in \mathbb{N}\}$ . Then instead of solving

approximately (3.12) we solve approximately the following system:

(5.1) 
$$A\Omega - \Omega \circ A = -S_J R_A$$
$$B\Omega - \Omega \circ B = -S_J R_B.$$

For a  $C^{\infty}$  function  $f = \sum_{n} \hat{f}_{n} \chi_{n}$  we define  $S_{J} f$  as

$$S_J f \stackrel{\text{def}}{=} \sum_{|n| < J} \hat{f}_n \chi_n.$$

Then the smoothing operators satisfy

(5.2) 
$$\|S_J f\|_{a+b} \le J^b \|f\|_a \\ \|S_J f\|_{C^{a+b}} \le J^{b+\sigma} \|f\|_{C^a}$$

where a > b > 0 and  $\sigma > N + 1$ . Also,

(5.3) 
$$\|(I - S_J)f\|_{a-b} \le J^{-b} \|f\|_a \\ \|(I - S_J)f\|_{C^{a-b}} \le CJ^{-b+\sigma} \|f\|_{C^a}$$

for  $a > b > \sigma > N + 1$ . These simple smoothing operators are convenient in our setting since they are well behaved with respect to the operator *L*. Namely, we have

$$L(S_J f, S_J g) = S_{\frac{J}{\xi}}(L(f, g)) + \mathcal{F}_{>\frac{J}{\xi}},$$

where  $\xi$  is a constant depending on *A* and *B* and the last term in the expression above consists of pieces of Fourier series for *f* and *g* involving only terms with  $|n| > \frac{J}{\xi}$ . This implies the following estimates:

(5.4) 
$$\|L(S_J f, S_J g)\|_a \leq \|S_{\frac{J}{\xi}}(L(f, g))\|_a + CJ^{-b}\|f, g\|_{a+b}$$
$$\|L(S_J f, S_J g)\|_{C^r} \leq \|S_{\frac{J}{\xi}}(L(f, g))\|_{C^{r+\sigma}} + CJ^{-b+\sigma}\|f, g\|_{C^{r+b}}$$
$$\leq J^{2\sigma}\|L(f, g)\|_{C^r} + CJ^{-b+\sigma}\|f, g\|_{C^{r+b}}$$

for any  $a > 0, b > \sigma > N + 1, r \ge 0$ .

5.2. Iterative step and the error estimate. At each step of the iterative scheme we first choose an appropriate smoothing operator  $S_J$ . In order to solve approximately (5.1) we use Lemma 4.6 to obtain the splitting

$$S_J R_A = \mathcal{P}(S_J R_A) + \mathcal{E}(S_J R_A)$$
$$S_J R_B = \mathcal{P}(S_J R_B) + \mathcal{E}(S_J R_B)$$

so that  $L(\mathcal{P}(S_J R_A), \mathcal{P}(S_J R_B)) = 0$ . Now from Lemma 4.4 the system

$$A\Omega - \Omega \circ A = -\mathcal{P}(S_J R_A)$$
$$B\Omega - \Omega \circ B = -\mathcal{P}(S_J R_B)$$

has an approximate solution  $\Omega$  such that

(5.5) 
$$\|\Omega\|_{C^r} \leq C_r \|\mathcal{P}(S_J R_A), \mathcal{P}(S_J R_B)\|_{C^{r+\sigma}}$$
$$\leq C_r \|S_J R_A, S_J R_B\|_{C^{r+2\sigma}}$$
$$\leq C_r J^{3\sigma} \|R_A, R_B\|_{C^r} \leq C_r J^{3\sigma} \|R\|_{C^r}$$

Here we used the estimates from Lemma 4.6 and the properties of smoothing (5.2). As mentioned in the Section 4.1,  $||R||_{C^r}$  stands for max{ $||R_A||_{C^r}$ ,  $||R_A||_{C^r}$ }.

The estimate (5.5) holds for any  $\sigma$  large enough:  $\sigma > \max\{M, M_0\}$ , so that the estimates in Lemma 4.4 and Lemma 4.6 hold. Then we form  $H \stackrel{\text{def}}{=} \text{id} + \Omega$  (since  $\Omega$  is made small in  $C^1$  throughout the iteration, H is invertible) and

$$\widetilde{\alpha}^{(1)} \stackrel{\text{def}}{=} H^{-1} \circ \widetilde{\alpha} \circ H$$

The new error is

$$R^{(1)} \stackrel{\text{def}}{=} \widetilde{\alpha}^{(1)} - \alpha$$

and it has two parts:

• the error coming from solving the linearized equation only approximately:

$$E_1 = \mathscr{C}(S_J R) + (I - S_J)R, \quad \text{and} \quad$$

• the standard error coming from the linearization

$$E_2 = \Omega \circ \widetilde{\alpha}^{(1)} - \Omega \circ \alpha + R \circ (\mathrm{id} + \Omega) - R.$$

*Estimate for E*<sub>1</sub>. Using Lemma 4.6 and the properties of smoothing (5.4), for every  $b > \sigma$  we have

$$\begin{aligned} \|\mathscr{E}(S_{J}R)\|_{C^{0}} &\leq C \|L(S_{J}R_{A}, S_{J}R_{B})\|_{C^{\sigma}} \\ &\leq C \|S_{\frac{J}{\xi}}L(R_{A}, R_{B})\|_{C^{2\sigma}} + CJ^{-b+\sigma} \|R\|_{C^{\sigma+b}} \\ &\leq C \left[J^{3\sigma}\|L(R_{A}, R_{B})\|_{C^{0}} + J^{-b+\sigma} \|R\|_{C^{\sigma+b}}\right]. \end{aligned}$$

Also, using (5.3),

$$||(I-S_J)R||_{C^0} \le C_l J^{-l+\sigma} ||R||_{C^1}$$

for any  $l \ge \sigma$ . Let  $b = l - \sigma$ . Thus, using Lemma 4.7, we have

(5.6) 
$$\|E_1\|_{C^0} \le CJ^{3\sigma} \|R\|_{C^0} \|R\|_{C^1} + CJ^{-l+2\sigma} \|R\|_{C^1}$$

for  $l > 2\sigma$ .

*Estimate for*  $E_2$ . The first part of  $E_2$  is estimated as follows:

$$\begin{split} \|\Omega \circ \widetilde{\alpha}^{(1)} - \Omega \circ \alpha\|_{C^0} &\leq C \|\Omega\|_{C^1} \|\widetilde{\alpha}^{(1)} - \alpha\|_{C^0} \\ &= C \|\Omega\|_{C^1} \|R^{(1)}\|_{C^0} \leq \frac{1}{4} \|R^{(1)}\|_{C^0}; \end{split}$$

thus this part is absorbed by  $||R^{(1)}||_{C^0}$  providing  $||\Omega||_{C^1}$  is bounded throughout the procedure. The second part of  $E_2$  we estimate by using (5.5)

(5.7) 
$$\|R(\mathrm{id} + \Omega) - R\|_{C^0} \le C \|R\|_{C^1} \|\Omega\|_{C^0} \le C J^{3\sigma} \|R\|_{C^1} \|R\|_{C^0}.$$

*Estimate of the new error*  $R^{(1)}$ . By combining (5.6) and (5.7) we obtain an estimate for the new error:

(5.8) 
$$\|R^{(1)}\|_{C^0} \le CJ^{3\sigma} \|R\|_{C^1} \|R\|_{C^0} + CJ^{-l+2\sigma} \|R\|_{C^l}$$

for any  $l > 2\sigma$ . From  $\tilde{\alpha}^{(1)} = H^{-1} \circ \tilde{\alpha} \circ H$  using the fact that  $\Omega$  is  $\mathbb{Z}^N$ -periodic and satisfies the estimate (5.5) we have

(5.9) 
$$\|R^{(1)}\|_{C^{l}} \leq C_{l}J^{3\sigma} \left(1 + \|R\|_{C^{l}}\right).$$

Therefore we have obtained what usually constitutes the basis for the iteration (see [38] for example), i.e., the "quadratic" estimate (5.8) for the low norm of the new error and the "linear" estimate (5.9) for some high norm of the new error with respect to the initial error R.

5.3. Setting up the iterative process. To set up the iterative process we first let

$$R^{(0)} = R, \qquad \widetilde{\alpha}^{(0)} = \widetilde{\alpha}, \qquad H^{(0)} = \mathrm{id}.$$

Now construct  $R^{(n)}$  inductively for every *n* for  $R^{(n)}$  choose an appropriate integer number  $J_n$  to obtain  $S_{J_n} R^{(n)}$  which produces, after solving approximately the linearized equation, new  $\Omega^{(n)}$ . Then define

(5.10) 
$$H^{(n)} = \mathrm{id} + \Omega^{(n)}$$
$$\widetilde{\alpha}^{(n+1)} = \left(H^{(n)}\right)^{-1} \circ \widetilde{\alpha}^{(n)} \circ H^{(n)}$$
$$R^{(n+1)} = \widetilde{\alpha}^{(n+1)} - \alpha.$$

Consequently,

$$\widetilde{\alpha}^{(n+1)} = \left(H^{(n)}\right)^{-1} \circ \left(H^{(n-1)}\right)^{-1} \circ \cdots \circ \left(H^{(0)}\right)^{-1} \circ \widetilde{\alpha} \circ H^{(0)} \circ \cdots \circ H^{(n)}$$
$$= \mathcal{H}_n^{-1} \circ \widetilde{\alpha} \circ \mathcal{H}_n$$

where  $\mathcal{H}_n \stackrel{\text{def}}{=} H^{(0)} \circ \cdots \circ H^{(n)}$ . To ensure the convergence of the process, set

(5.11) 
$$\|R^{(n)}\|_{C^{0}} < \varepsilon_{n} = \varepsilon^{(k^{n})}$$
$$\|R^{(n)}\|_{C^{1}} < \varepsilon_{n}^{-1}$$
$$\|\Omega^{(n)}\|_{C^{1}} < \varepsilon_{n}^{1/2}$$
$$J_{n} = \varepsilon_{n}^{-\frac{3(3\sigma+2)}{3(3\sigma+2)}}$$

where  $k = \frac{4}{3}$ . At this point, fix *l*:

(5.12) 
$$l = 23\sigma + 15,$$

where  $\sigma = \sigma(A, B, N) = \max\{M, M_0\}$  is a constant for which the estimates (4.60) and (4.27) hold. The constant *l* is chosen so that the process converges and the convergence is proved in the subsequent section.

5.4. *Convergence*. By induction it is proved that all the bounds (5.11) hold for every  $n \in \mathbb{N}$ :

$$\|R^{(n+1)}\|_{C^{l}} \leq C_{l} J_{n}^{3\sigma} \left(1 + \|R^{(n)}\|_{C^{l}}\right) \leq C_{l} J_{n}^{3\sigma} \left(1 + \varepsilon_{n}^{-1}\right)$$
$$\leq 2C_{l} J_{n}^{3\sigma} \varepsilon_{n}^{-1} \leq 2C_{l} \varepsilon_{n}^{-\frac{3\sigma}{3(3\sigma+2)}} \varepsilon_{n}^{-1} < \varepsilon_{n}^{-\frac{1}{3}-1} = \varepsilon_{n}^{-\frac{4}{3}}$$
$$= (\varepsilon_{n+1})^{-1}.$$

From interpolation inequalities it follows that

$$\|R^{(n)}\|_{C^1} \leq C_l \|R^{(n)}\|_{C^0}^{1-\frac{1}{l}} \|R^{(n)}\|_{C^l}^{\frac{1}{l}}.$$

Along with (5.8) this implies

$$\begin{aligned} \|R^{(n+1)}\|_{C^0} &\leq C \left[ \varepsilon_n^{-\frac{3\sigma}{3(3\sigma+2)}} \varepsilon_n^{1-\frac{1}{l}} \varepsilon_n^{-\frac{1}{l}} \varepsilon_n + \varepsilon_n^{\frac{l-2\sigma}{3(3\sigma+2)}} \varepsilon_n^{-1} \right] \\ &= C \left[ \varepsilon_n^{-\frac{\sigma}{3\sigma+2}+2(1-\frac{1}{l})} + \varepsilon_n^{\frac{l-2\sigma}{3(3\sigma+2)}-1} \right] = C \left[ \varepsilon_n^x + \varepsilon_n^y \right] \\ &\leq \varepsilon_n^{\frac{4}{3}} = \varepsilon_{n+1} \end{aligned}$$

providing  $x > \frac{4}{3}$ ,  $y > \frac{4}{3}$ ; i.e.,

$$x = -\frac{\sigma}{3\sigma + 2} + 2\left(1 - \frac{1}{l}\right) > \frac{4}{3}$$
$$y = \frac{l - 2\sigma}{3(3\sigma + 2)} - 1 > \frac{4}{3}.$$

Both inequalities above are satisfied for  $l \ge 23\sigma + 15$ . We note here that with more precision (by changing the rate of convergence) the constant l can be made somewhat smaller (see Section 7.1). Using (5.5) we may check the  $C^1$  bound for  $\Omega$ :

$$\|\Omega^{(n+1)}\|_{C^{1}} \leq C \cdot J_{n}^{3\sigma+1} \|R^{(n)}\|_{C^{0}} \leq C \cdot J_{n}^{3\sigma+1} \varepsilon_{n} = C \cdot \varepsilon_{n}^{-\frac{3\sigma+1}{3(3\sigma+2)}+1} < \varepsilon_{n}^{\frac{2}{3}} = \varepsilon_{n+1}^{\frac{1}{2}}.$$

Thus for sufficiently small  $||R||_{C^0}$  and  $||R||_{C^1}$  the process converges to a solution  $\Omega \in C^1$  with  $||\Omega||_{\mathscr{C}^1} < \frac{1}{4}$ .

Now using interpolation inequalities, the fact that the process converges in any  $C^r$  norm follows easily just as in [31] or as in [38]. We repeat this argument here for completeness. For arbitrary *m* from (5.9) we have

$$\|R^{(n+1)}\|_{C^m} \leq C_m J_n^{3\sigma} \left(1 + \|R^{(n)}\|_{C^m}\right) \leq \varepsilon_n^{-\frac{1}{3}} \left(1 + \|R^{(n)}\|_{C^m}\right),\\ \|R^{(n)}\|_{C^m} \leq C_m \prod_{\nu=1}^{n-1} \varepsilon_{\nu}^{-\frac{1}{3}} \left(1 + \|R\|_{C^m}\right) \leq \varepsilon_n^{-1} C_m,$$

where in the second line above the constant  $C_m$  is  $C_m := C_m (1 + ||R||_{C^m})$ . Let m = 3k. Then

$$\|R^{(n)}\|_{C^{k}} \leq C_{k} \|R^{(n)}\|_{C^{0}}^{\frac{2}{3}} \|R^{(n)}\|_{C^{3k}}^{\frac{1}{3}} < C_{k} \varepsilon_{n}^{\frac{2}{3}} \varepsilon_{n}^{-\frac{1}{3}} = C_{k} \varepsilon_{n}^{\frac{1}{3}},$$
  
$$\|\Omega^{(n)}\|_{C^{k}} \leq C_{k} J_{n}^{3\sigma} \|R^{(n)}\|_{C^{k}} \leq C_{k} \varepsilon_{n}^{-\frac{3\sigma}{3(3\sigma+2)}} \varepsilon_{n}^{\frac{1}{3}} = C_{k} \varepsilon_{n}^{\delta},$$

with  $\delta = \frac{2}{3(3\sigma+2)} > 0$  and the constant  $C_k$  changing throughout the above procedure, but it depends only on k and the  $C^{3k}$  norm of the initial perturbation R.

This implies the convergence of the sequence  $\mathcal{H}_n$  in  $C^k$  norm for every  $k \in \mathbb{N}$ , i.e., the limit  $\mathcal{H}$  is a  $C^{\infty}$  map.

## 6. Proof of Theorem 2

Definition 2. A  $C^{\infty}$  foliation  $\mathcal{F}$  of a manifold M is  $C^{\infty,l,\infty}$  locally rigid if for any sufficiently  $C^l$  close  $C^{\infty}$  foliation  $\mathcal{F}'$  there exists a  $C^{\infty}$  diffeomorphism H of M taking leaves of  $\mathcal{F}$  to leaves of  $\mathcal{F}'$ .

Definition 3. An action  $\alpha$  of  $\mathbb{R}^k$  on M is  $C^{\infty,l,\infty}$  orbit foliation rigid if its orbit foliation is  $C^{\infty,l,\infty}$  locally rigid.

The following lemma is completely general and quite possibly may be found in the literature. Since we are not aware of a specific reference we provide a proof.

LEMMA 6.1.  $C^{\infty,l,\infty}$  local rigidity for an action  $\gamma$  of  $\mathbb{Z}^k$  by diffeomorphisms on a manifold M implies  $C^{\infty,l,\infty}$  orbit foliation rigidity for the suspension action  $\alpha(\gamma)$  on the suspension manifold  $N(\gamma)$ .

*Proof.* Let  $\Gamma: \mathbb{Z}^k \to \text{Diff}^{\infty}(M)$  be a representation of  $\mathbb{Z}^k$  defining a  $\mathbb{Z}^k$  action  $\gamma$  by diffeomorphisms of M. Let  $\alpha(\gamma): \mathbb{R}^k \times N(\gamma) \to N(\gamma)$  be the corresponding suspension action with the orbit foliation  $\mathcal{F}(\gamma)$  and let  $\tilde{\alpha}$  be a  $C^l$  small  $C^{\infty}$  perturbation of  $\alpha(\gamma)$  with the orbit foliation  $\mathcal{F}(\gamma)$  which is  $C^l$  close to  $\mathcal{F}(\gamma)$ .

If the perturbation is sufficiently small the orbit foliation of the perturbation is still transversal to the fibers M of the suspension manifold  $N(\gamma)$ . Let  $\tilde{\Gamma}: \mathbb{Z}^k \to$  $\text{Diff}^{\infty}(M)$  be the holonomy of the perturbed foliation  $\overline{\mathscr{F}(\gamma)}$ , defining an action  $\tilde{\gamma}$ of  $\mathbb{Z}^k$  on M. Then there exists a  $C^{\infty}$  diffeomorphism  $H_1: N(\tilde{\gamma}) \to N(\gamma)$  taking leaves of the foliation  $\mathscr{F}(\tilde{\gamma})$  of the suspension action over  $\tilde{\gamma}$  to the leaves of  $\overline{\mathscr{F}(\gamma)}$ (Theorem 3 in [3, §5.4]).

On the other hand, since  $\gamma$  and  $\tilde{\gamma}$  are sufficiently  $C^l$  close,  $C^{\infty} \mathbb{Z}^k$  actions there is a  $C^{\infty}$  conjugacy  $h: M \to M$  such that  $h \circ \gamma = \tilde{\gamma} \circ h$  according to our assumption on the local rigidity of  $\gamma$ . This implies the existence of a  $C^{\infty}$  orbit equivalence for the corresponding suspensions, i.e., there exists  $H_2: N(\gamma) \to N(\tilde{\gamma})$ taking leaves of  $\mathcal{F}(\gamma)$  to leaves of  $\mathcal{F}(\tilde{\gamma})$  (Theorem 2 in [3, §5.4]).

Thus the  $C^{\infty}$  diffeomorphism  $H = H_1 \circ H_2$  on  $N(\gamma)$  is an orbit equivalence for actions  $\alpha(\gamma)$  and  $\tilde{\alpha}$  which implies  $C^{\infty,l,\infty}$  orbit foliation rigidity for  $\alpha(\gamma)$ .  $\Box$  It follows from Lemma 6.1 and Theorem 1 that for a given suspension action and its small perturbation there exists a  $C^{\infty}$  orbit equivalence taking orbits of the original to the orbits of the perturbed action. This reduces the question of local rigidity of suspensions to considering small perturbations along the leaves of orbit foliation only. However such perturbations are given by  $\mathbb{R}^k$  valued cocycles over  $\alpha(\gamma)$ .

It is proved in [23, §4.2] that  $C^{\infty}$  cocycle rigidity of a  $\mathbb{Z}^k$  action  $\gamma$  by toral automorphisms implies  $C^{\infty}$  cocycle rigidity for suspension  $\alpha(\gamma)$ . For the proof of  $C^{\infty}$  cocycle rigidity of a  $\mathbb{Z}^k$  action  $\gamma$  with  $k \ge 2$  and all nontrivial elements of the ergodic action, we refer to [24] or [18]. (The proof of this fact is also contained in part (i) of the proof of Lemma 4.4 for untwisted cocycles.) Thus  $\alpha(\gamma)$  is  $C^{\infty}$  cocycle rigid, which implies that perturbations in orbit direction are conjugate to the original action up to an automorphism of  $\mathbb{R}^k$ . This completes the proof of Theorem 2.

# 7. Comments on finitely differentiable and analytic case

7.1. *Finitely differentiable case.* It is clear that all the arguments in the proof of Theorem 1 only use a finite number of derivatives so that our rigidity results in a modified form hold when the perturbed action  $\alpha$  is only finitely differentiable. We discuss now specific modifications which appear this way.

Let *R* be  $C^m$ . Then the estimate (4.10) for the approximate solution of the linearized equation obtained in Lemma 4.4 still holds if  $m > \sigma$ , where  $\sigma > M_0$  and  $M_0$  is defined in (4.27). Similarly, the estimates (4.45) and (4.46) obtained in Lemma 4.6 which are used later in the proof of the Theorem 1 (Section 5) hold when *R* is  $C^m$  and m > M, where *M* is defined in (4.60). Now the iterative proof in Section 5 applies in the following setting.

Let  $\alpha$  be a  $\mathbb{Z}^k$  action by automorphisms of  $\mathbb{T}^N$  as in the statement of Theorem 1. Let  $\sigma \in \mathbb{N}$  be a fixed number greater than constants M (in (4.60)) and  $M_0$  (in (4.27)) defined in Lemmas 4.4 and 4.6.

The convergence set-up is the same as in the Section 5.3, the modification only comes in determining the speed of the convergence. Instead of (5.11) now we set

(7.1)  
$$\|R^{(n)}\|_{C^{0}} < \varepsilon_{n} = \varepsilon^{(k^{n})},$$
$$\|R^{(n)}\|_{C^{1}} < \varepsilon_{n}^{-1},$$
$$\|\Omega^{(n)}\|_{C^{1}} < \varepsilon_{n}^{\frac{2-k}{k}},$$
$$J_{n} = \varepsilon_{n}^{-\frac{k-1}{(3\sigma+2)}}.$$

where 1 < k < 2. While making sure that these bounds hold for every *n*, we will obtain a lower bound on *l* which depends on the speed of convergence *k*.

We first check the bound for the  $C^{l}$  norm using the estimate (5.9):

(7.2) 
$$\|R^{(n+1)}\|_{C^{l}} \le C_{l} J_{n}^{3\sigma} (1 + \|R^{(n)}\|_{C_{l}}) \le C_{l}^{\prime} \varepsilon_{n}^{-(k-1)} \varepsilon_{n}^{-1} = \varepsilon_{n+1}^{-1}.$$

Using (5.5) it is easy to check the bound for  $\Omega$ :

(7.3) 
$$\|\Omega^{(n)}\|_{C^1} \le C \varepsilon_n^{-\frac{k-1}{3\sigma+2}(3\sigma+1)} \varepsilon_n \le \varepsilon_n^{2-k} \le \varepsilon_{n+1}^{\frac{2-k}{k}}$$

Now for the  $C^0$  bound we have, as in Section 5.4, using (5.8) and the interpolation estimates:

$$\begin{split} \|R^{(n+1)}\|_{C^{l}} &\leq C_{l} J_{n}^{3\sigma} \|R^{(n)}\|_{C^{1}} \|R^{(n)}\|_{C^{0}} + C_{l} J_{n}^{-l+2\sigma} \|R^{(n)}\|_{C^{l}} \\ &\leq C_{l} J_{n}^{3\sigma} \|R^{(n)}\|_{C^{0}}^{2-\frac{1}{l}} \|R^{(n)}\|_{C^{l}}^{\frac{1}{l}} + C_{l} J_{n}^{-l+2\sigma} \|R^{(n)}\|_{C^{l}} \\ &\leq C_{l} \varepsilon_{n}^{-\frac{3\sigma(k_{1})}{3\sigma+2}} \varepsilon_{n}^{2-\frac{1}{l}} \varepsilon_{n}^{-\frac{1}{l}} + \varepsilon_{n}^{-\frac{(k-1)(-l+2\sigma)}{3\sigma+2}} \varepsilon_{n}^{-1} \\ &\leq \varepsilon_{n}^{-k+3-\frac{2}{l}} + \varepsilon_{n}^{\frac{(k-1)(l-2\sigma)}{3\sigma+2}-1}. \end{split}$$

Thus if  $-k + 3 - \frac{2}{l} > k$  and  $\frac{(k-1)(l-2\sigma)}{3\sigma+2} - 1 > k$  then we have  $||R^{(n+1)}||_{C^l} \le \varepsilon_{n+1}$ . The first condition gives  $l > \frac{2}{3-2k}$  and the second one  $l > 2\sigma + \frac{k+1}{k-1}(3\sigma+2)$ . The first condition is satisfied already for l > 2. The second one actually gives the dependence on k. Namely, as the speed of convergence k approaches 2, the lower bound for l approaches  $l_0 := 11\sigma + 6$ . Thus if one chooses  $l = l_0 + \delta$  for  $\delta > 0$ , then it is possible to choose the rate of convergence  $k = k(\delta)$  so that all the bounds (7.1) hold for every n.

This implies the convergence of the procedure in  $C^1$  norm. Therefore, the perturbation needs to be close to the initial action in  $C^l$  with l (only) strictly larger than  $l_0$ , in order to obtain a  $C^1$  solution.

To obtain more derivatives for the solution, we need to assume that the perturbation is more regular than  $C^{l_0+\delta}$ . Again, at this point we use interpolation inequalities. Assume that the initial perturbation is  $C^m$  and let 1 < r < m. As in Section 5.4 it is easy to check that one has  $||R^{(n)}||_{C^m} \leq C_m \varepsilon_n^{-1} ||R^{(0)}||_{C^m}$ . Therefore we have

$$\|R^{(n)}\|_{C^{r}} \leq C \|R^{(n)}\|_{C^{0}}^{1-\frac{r}{m}} \|R^{(n)}\|_{C^{m}}^{\frac{r}{m}} < C_{m}\varepsilon_{n}^{1-2\frac{r}{m}}$$
$$\|\Omega^{(n)}\|_{C^{r}} \leq C J_{n}^{3\sigma} \|R^{(n)}\|_{C^{r}} \leq C_{m}\varepsilon_{n}^{-(k-1)\frac{3\sigma}{(3\sigma+2)}}\varepsilon_{n}^{1-2\frac{r}{m}}$$

Therefore, in order to obtain a solution of order r, we need that  $\frac{r}{m} < 1 - \frac{k}{2}$ . So if the perturbation is  $C^{\infty}$  then for any  $l > l_0$  the solution is  $C^{\infty}$  with k chosen close to 2 if l is close to  $l_0$  as described above.

Now if the perturbation is only  $C^m$  and close to the unperturbed action in  $C^l$  for some  $l > l_0$ , then choose k so that  $l > 2\sigma + \frac{k+1}{k-1}(3\sigma + 2)$ , that is choose k close to but larger than  $k_0(l) = \frac{l+5\sigma+2}{l-3\sigma-2} \in (1,2)$ . Then the solution is  $C^r$  for every  $r < (1 - \frac{k_0(l)}{2})m$ .

7.2. Analytic case. Now suppose the perturbed action  $\alpha$  is real analytic. A natural question is whether the unique  $C^{\infty}$  conjugacy with the linear action is also analytic.

Both estimates in Lemma 4.2 and in part d) in Section 4.3 can be obtained in the category of analytic spaces  $A_{\sigma}$  allowing arbitrary small loss of domain  $\delta$ in all directions and assuming that there is no growth for the action in the neutral direction.

The algebraic part of the proof of Lemma 4.5 holds too, thus one can define maps  $\mathscr{E}\psi$ ,  $\mathscr{E}\theta$ ,  $\mathscr{P}\psi$ , and  $\mathscr{P}\theta$ . The analytic estimates for the  $\mathscr{E}\psi$  and  $\mathscr{P}\psi$  with small loss of domain hold (for estimating  $\mathscr{P}\psi$  we need to use that  $\theta$  and  $\psi$  are small with respect to  $L(\theta, \psi)$  and thus quadratically small with respect to  $\psi$  which follows by using Cauchy estimates for  $L(\theta, \psi)$ ). The problem is that the estimates for other functions,  $\mathscr{E}\theta$  and  $\mathscr{P}\theta$ , can be obtained by this method only in a domain which is in some directions much smaller than the initial one (multiplied by a constant less than one coming from the contraction of some directions by the action of A). Such a loss at every iterative step cannot lead to a convergence. Thus, lack of analytic result may be due to specific constructions in Section 4.5 but it may also be due to the highly hyperbolic nature of the problem at hand.

# 8. On the application of general implicit function theorems

The iterative procedure carried out in Sections 5.3 and 5.4 relies essentially on the result of Lemma 4.6. Taking into account Lemmas 4.4 and 4.7, the result of Lemma 4.6 can be interpreted as *there exists an approximate right inverse for the operator* 

$$T_0: \Omega \mapsto \alpha \, \Omega - \Omega \circ \alpha$$

in the  $C^{\infty}$  category, with tame estimates for the  $C^r$  norms. Operator  $T_0$  maps  $C^{\infty}$  maps on the torus to  $C^{\infty} \alpha$ -twisted cocycles over the action  $\alpha$ . (Note. As before  $\alpha$  denotes here the unperturbed linear  $\mathbb{Z}^k$  action on the torus  $\mathbb{T}^N$  by ergodic automorphisms.) Now, operator  $T_0$  is simply the linearization of the conjugacy operator

$$\mathcal{T}: H \to H^{-1} \circ \alpha \circ H$$

which takes the space of  $C^{\infty}$  diffeomorphisms of  $\mathbb{T}^N$  to the space  $\operatorname{Act}^{\infty}(\mathbb{Z}^k, \mathbb{T}^N)$  of smooth actions. Thus what we needed for the convergence in Section 5.4 is *only* an approximate right inverse of the linearization of  $\mathcal{T}$  at the identity. The identity is in our set-up the *initial guess* for the solution. Let us denote by  $T_H$  the linearization of  $\mathcal{T}$  at H when H is not the identity.

It is obvious that the problem of finding a conjugacy for an arbitrary small perturbation of an action is a problem of inverting a nonlinear operator, and it can also be viewed as an implicit function problem.

After the first results of Kolmogorov, Arnold and Moser where the method of linearization and successive iterations was applied to produce a solution to a nonlinear problem, there were several generalizations and refinements of those iterative schemes. These results are labeled in literature as "hard implicit function theorems" or "generalized implicit function theorems". Typically, these results state that under certain conditions on the linearization of the nonlinear operator *in some neighborhood of the initial guess* (most importantly, assuming existence of

an approximate or an exact right inverse of the linearization) there exists a solution to the nonlinear problem.

Below we discuss two such results and we discuss why they can be applied to the problem treated in Theorem 1, or not.

One of early generalized implicit function theorems is due to Zehnder in 1975 [38, Theorem 3.1]. The main requirement for the application of Zehnder's theorem in the set-up of Theorem 1 is existence of an *approximate* right inverse of the linearization  $T_H$  for H in a neighborhood of the initial guess. However, his condition on how approximate should the right inverse be at the initial guess, is too strong for our purposes: it requires an exact right inverse at the initial guess. This is clearly something we cannot produce in our situation: existence of an exact right inverse at the identity would imply that the space of perturbations lifts to a space of  $\alpha$  twisted cocycles, which is not true. (As explained in Section 3.3 the linearization produces only almost  $\alpha$  twisted cocycles, not actual  $\alpha$  twisted cocycles.)

On the other hand, at the end of his paper Zehnder makes a very important remark that due to the algebraic structure of conjugacy classes, the main requirement of his result concerning existence of an approximate right inverse in a neighborhood, can be relaxed to existence of an approximate right inverse of the linearization operator *at the initial guess only*. However, this is not proved explicitly in [38] and it is not clear from the remark what would be exactly the modifications in the result [38, Theorem 3.1] if one restricts it to conjugacy operator. It is possible that Zehnder's result can be adapted in this direction, so that it can be applied in the set-up of Theorem 1 to substitute for explicit proof of convergence in Section 5.4. This would only slightly improve the result in the finitely differentiable case due to the more optimal use of smoothing operator's in Zehnder's work, but it would result in considerable divergence from the main issue of the current paper.

The second result we wish to discuss applies in the set-up of Theorem 1, although it does not apply in the finitely differentiable set-up. This is the Nash-Moser theorem for exact sequences due to Hamilton [15, Theorem 3.1.1] in 1982, an extremely convenient result when one is dealing with rigidity of group actions. We refer to [12] for an excellent account on application of Nash-Moser theorem for exact sequences to rigidity of actions of finitely generated finitely presented groups, and we restrict here just to few remarks relative to the current paper.

The commutativity relation in the acting group induces an operator  $\mathcal{M}$  which takes a pair of diffeomorphisms to their commutator. Let  $M_{\mathcal{T}(H)}$  be the linearization of  $\mathcal{M}$  at  $\mathcal{T}(H)$ , and let  $M_0$  be the linearization of  $\mathcal{M}$  at  $\alpha = \mathcal{T}(id)$ . The main requirement for the application of Hamilton's theorem is existence of a splitting of the short exact sequence of operators  $M_{\mathcal{T}(H)}T_H = 0$  in the neighborhood of the initial guess. In the current paper Lemma 4.4 shows that  $M_0T_0 = 0$  indeed holds, and Lemma 4.6 shows the existence of the splitting for the short exact sequence  $M_0T_0 = 0$ . (The operator  $M_0$  restricted to two generators is the operator L which is discussed in Section 4.)

Zehnder's remark that the algebraic structure of conjugacy problems relaxes the conditions of his theorem, has been formalized and proved in the context of Hamilton's result by Benveniste [1, Lemma 4.3] in 2000. Namely, [1, Lemma 4.3] shows that if the nonlinear problem under consideration is the conjugacy problem, then existence of a tame splitting of  $M_0T_0 = 0$  implies existence of a tame splitting of  $M_{\mathcal{T}(H)}T_H = 0$  for H in some small neighborhood of the identity. Benveniste applied this fact along with the Hamilton's result to prove local rigidity for isometric actions by lattices in simple Lie groups. More recently similar approach was used by Fisher [12] to prove local rigidity for isometric actions of discrete groups with property (T). Lemma 4.3 in [1], combined with Lemmas 4.6, 4.4 and 4.7, imply that Hamilton's theorem can be applied in the set-up of Theorem 1 to give a  $C^{\infty}$  conjugacy. However, Hamilton's result does not give explicitly the topology in which the perturbation has to be small nor how it depends on the unperturbed action and it does not apply to the finitely differentiable situation.

# 9. Existence of genuinely partially hyperbolic actions

To prove the statement of Theorem 3 we first give a proof that there are no irreducible automorphisms in odd dimensions with nontrivial neutral direction. This is proved in [33], but the proof we give here is considerably simpler. Then we eliminate the possibility of having genuinely partially hyperbolic actions on  $\mathbb{T}^2$ and  $\mathbb{T}^4$ . This leaves open the question of existence of examples on  $\mathbb{T}^N$  for  $N \ge 6$ . We then give an outline of an *explicit construction* of an irreducible example of a genuinely partially hyperbolic action on  $\mathbb{T}^6$  with two-dimensional neutral direction. This construction can be used further as a model for constructing examples with various properties (with neutral direction of any even dimension, for example). Finally, we show that there exist irreducible examples of genuinely partially hyperbolic actions in any even dimension  $N \ge 6$ . By combining these results we obtain reducible examples in any odd dimension  $N \ge 9$ .

9.1. There are no irreducible examples of genuinely partially hyperbolic actions on  $\mathbb{T}^N$  for N odd. As mentioned in the introduction to this section, there are no irreducible genuinely partially hyperbolic toral automorphisms in any odd dimension, thus there are no such actions in any odd dimension either. This was proved in [33]. This fact also follows from a simple number theoretic argument. Namely, if we let p(x) be the irreducible characteristic polynomial of an integer matrix A of degree N, let  $\psi$  be root of p of absolute value 1, and let  $\theta \stackrel{\text{def}}{=} \psi + \psi^{-1}$ , then for the corresponding number fields  $L = \mathbb{Q}(\psi)$  and  $K = \mathbb{Q}(\theta)$  we have

$$|L:K||K:\mathbb{Q}|=N.$$

On the other hand *K* is real since  $\theta = \psi + \psi^{-1} = \psi + \overline{\psi} \in \mathbb{R}$  so  $|L:K| \ge 2$ . But we also have that  $\psi^2 - \psi\theta + 1 = 0$ ; therefore  $|L:K| \le 2$ . This implies that |L:K| = 2; thus *N* has to be even. 9.2. There are no irreducible examples of genuinely partially hyperbolic actions on  $\mathbb{T}^N$  for N = 2 or N = 4.

Case N = 2. An integer  $2 \times 2$  matrix which induces an ergodic toral automorphism must have a nontrivial expanding and a nontrivial contraction direction. Since it is of dimension 2, there are no other directions, in particular, it cannot have any eigenvalues on the unit circle. We note that it is, of course, possible to have one hyperbolic integer matrix of determinant one in dimension 2, but no two commuting ones (that are not powers of each other) exist due to the Dirichlet Units Theorem [19].

Case N = 4. In dimension 4 it is possible to have one matrix with desired properties, namely a matrix that is integer and has two complex conjugate eigenvalues of absolute value one, and two real eigenvalues  $\lambda$  and  $\lambda^{-1}$ . To produce such matrix it is enough to choose a quadratic irreducible polynomial p(x) with two real eigenvalues one bigger than 2 and the other less than 2 in absolute value. Then the substitution  $x^2 p(x + \frac{1}{x})$  gives a fourth degree polynomial which is a characteristic polynomial of a matrix with desired properties. However, it is not possible to have two commuting matrices with the properties above and with common neutral subspace (this last one is a necessary requirement for otherwise the action generated by the two matrices would be hyperbolic). Indeed, if B commutes with A and has real eigenvalues  $\mu$  and  $\mu^{-1}$ , then, because of irreducibility requirement, both  $\lambda$ and  $\mu$  are irrational, thus we can choose an integer vector (l, k) in  $\mathbb{Z}^2$  such that  $\lambda^{l} \mu^{k}$  is close to 1. But then the same holds true for the remaining real eigenvalues. Moreover, rotations in the neutral direction are assumed irrational; therefore, by choosing k and l large enough, they can be made close to one, also. This implies that some power of matrix  $A^{l}B^{k}$  is integer matrix close to identity. This is not possible unless the matrix is identity itself, in which case the action has a nontrivial nonergodic element, thus would not be a higher rank action.

9.3. Construction of an irreducible partially hyperbolic action with two-dimensional neutral direction in dimension 6. The following example of irreducible genuinely partially hyperbolic  $\mathbb{Z}^2$  action on the torus  $\mathbb{T}^6$  was produced by S. Katok with the use of PARI program. For the background on use of algebraic number theory for producing examples of higher rank actions by toral automorphisms we refer to [19].

Starting with an irreducible cubic polynomial  $x^3 - 2x^2 - 8x + 1$  with two real roots of absolute value larger than 2 and one less than 2 in absolute value, by substituting  $x \rightarrow x + \frac{1}{x}$  we obtain an irreducible recurrent polynomial

$$f(x) = x^{6} - 2x^{5} - 5x^{4} - 3x^{3} - 5x^{2} - 2x + 1,$$

which has four real roots and a pair of complex conjugate roots. The complex conjugate roots have to be of absolute value 1.

Let  $K = \mathbb{Q}(\lambda)$  be the number field corresponding to f(x). The fundamental units are  $\lambda_1 = \lambda$ ,  $\lambda_2 = \lambda^5 - 3\lambda^4 - 2\lambda^3 - \lambda^2 - 4\lambda + 1$ ,  $\lambda_3 = \lambda^4 - 2\lambda^3 - 6\lambda^2 - \lambda + 1$ ,  $\lambda_4 = 2\lambda^5 - 6\lambda^4 - 3\lambda^3 - 6\lambda^2 - 6\lambda$ . It turns out that  $\lambda_4$  produces a matrix with same neutral subspace as A.

With respect to the basis  $\{1, \lambda, \lambda^2, \lambda^3, \lambda^4, \lambda^5\}$  in  $\mathbb{Z}[\lambda]$  multiplications by  $\lambda_1$  and  $\lambda_4$  are given by the matrices

$$A = \begin{pmatrix} 0 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \\ -1 & 2 & 5 & 3 & 5 & 2 \end{pmatrix},$$
$$B = \begin{pmatrix} 0 & -6 & -6 & -3 & -6 & 2 \\ -2 & 4 & 4 & 0 & 7 & -2 \\ 2 & -6 & -6 & -2 & -10 & 3 \\ -3 & 8 & 9 & 3 & 13 & -4 \\ 4 & -11 & -12 & -3 & -17 & 5 \\ -5 & 14 & 14 & 3 & 22 & -7 \end{pmatrix}$$

Since  $B = 2A^5 - 6A^4 - 3A^3 - 6A^2 - 6A$ , AB = BA, and since the minimal polynomial of the matrix *B* is recurrent,  $x^6 + 23x^5 + 16x^4 - 60x^3 + 16x^2 + 23x + 1$  and also has four real roots, the two complex conjugate roots are of absolute value 1.

9.4. Existence of irreducible examples of genuinely partially hyperbolic actions in any even dimension greater than 6. In this section we use the Dirichlet units theorem to prove the main part of Theorem 3 stated in Section 1.3, that is, to produce examples of  $\mathbb{Z}^2$  genuinely partially hyperbolic actions in any even dimension  $N \ge 6$ .

Let q(x) be an irreducible integer polynomial of degree *n* with all real roots. Let  $r(x) = q(x + \frac{1}{x})x^n$  be the recurrent polynomial of degree 2n given by *q*. Let  $L = \mathbb{Q}(\psi)$  and  $K = \mathbb{Q}(\theta)$  be the number fields corresponding to *q* and *r* respectively with  $\psi + \psi^{-1} = \theta$ . Let  $\sigma_i$  be embeddings of *K* into  $\mathbb{C}$ . Since all roots of *r* come in pairs, let  $\sigma_i(\psi)\sigma_{n+i}(\psi) = 1$  for all i = 1, ..., n. Let  $\alpha = p(\psi) \in L$  with deg $(p) \leq 2n$  be any other element in *L*. Then  $\sigma_i(p(\psi^{-1})) = p(\sigma_i(\psi^{-1})) = p((\sigma_i(\psi))^{-1}) = p(\sigma_{n+i}(\psi)) = \sigma_{n+i}(p(\psi))$ . This implies that  $p(\psi)$  and  $p(\psi^{-1})$  have same norms  $N(p(\psi)) = N(p(\psi^{-1}) = \sigma_i(\psi) \cdots \sigma_{2n}(\psi)$ . Therefore,  $p(\psi)$  is a unit if and only if  $p(\psi^{-1})$  is.

Let  $U_L$  and  $U_K$  be groups of units of L and K, respectively. Define a homomorphism  $f: U_L \to U_K$  by

$$f(p(\psi)) \stackrel{\text{def}}{=} p(\psi) p(\psi^{-1}).$$

We first show that  $f(U_L)$  is indeed in  $U_K$ . Since for any integer polynomial p(x), we have  $p(x)p(x^{-1}) = P(x + x^{-1})$  where P is a rational polynomial, it follows that  $f(p(\psi)) = P(\theta)$  thus  $f(U_L) \subset K$ . Also  $p(\psi)$  being a unit implies that  $p(\psi^{-1})$  is also a unit and thus  $P(\theta)$  is a unit in  $U_L$ . Since  $\psi^2 - \theta\psi + 1 = 0$  we have  $|L:K| \le 2$ . With  $|L:L \cap \mathbb{R}| = 2$  and the fact that K is real, this implies  $K = L \cap \mathbb{R}$ . Therefore, a real unit in  $U_L$  must lie in  $U_K$ . This proves that the image of f is in  $U_K$ . It is then easy to check that f is a group homeomorphism.

In order to obtain matrices that commute with matrix A whose characteristic polynomial is r(x) it is enough to show that the kernel of f contains at least two independent units. We show that the kernel contains s independent units where s is the number of roots of q(x) of absolute value bigger than 2 (which implies 2s real roots for r(x)). The rest t = n - s roots of q(x) are of absolute value less than 2 and they induce t pairs of complex conjugate roots of r(x) of absolute value 1. By Dirichlet units theorem the structure of groups  $U_L$  and  $U_K$  is known. Namely, every unit in  $U_L$  can be expressed as  $\rho u_1^{j_1} u_2^{j_2} \cdots u_{s+n-1}^{j_{s+n-1}}$ , where  $u_i$  are independent units. The images  $f(u_i)$  of fundamental units of  $U_L$  are units in  $U_K$ . By Dirichlet units theorem  $U_K$  can have at most n-1 independent units. Therefore, in the set  $S = \{f(u_i) | i = 1, \ldots, n+s-1\}$  there can be at most n-1 independent units, without loss of generality assume those are  $f(u_i), i = 1, \ldots, n-1$ . This implies that there are at least s relations of the kind:

$$1 = f(u_1)^{j_{1,k}} f(u_2)^{j_{2,k}} \cdots f(u_{n-1})^{j_{n-1,k}} f(u_k)^{j_k},$$

where k = n, ..., n + s. Since f is a homomorphism, this implies that there are s units in the kernel of f:

$$v_k = u_1^{j_{1,k}} u_2^{j_{2,k}} \cdots u_{n-1}^{j_{n-1,k}} u_k^{j_k}$$

for k = n, ..., n + s. Moreover, all  $v_k$  are independent since  $u_i$  are.

This implies that if  $s \ge 2$  then there exists a unit  $v = p(\psi)$  such that B = p(A) commutes with A, A and B are multiplicatively independent, and B has the same neutral eigenspace as A. Namely, since v is in the kernel of f we have  $\sigma_i(f(v)) = 1$  for all i, i.e.,  $1 = \sigma_i(p(\psi))\sigma_i(p(\psi^{-1}))p(\sigma_i(\psi))p(\sigma_i(\psi^{-1})) = p(\psi_i)p(\psi_i^{-1})$ .

In particular, if  $\psi_i$  is a root of r of absolute value 1, then  $\psi_i^{-1} = \overline{\psi_i}$ ; therefore  $1 = p(\psi_i)p(\overline{\psi_i}) = p(\psi_i)\overline{p(\psi_i)} = |p(\psi_i)|$ . Thus B has neutral eigenvalues in the same direction as A and the action generated by A and B is genuinely partially hyperbolic.

9.5. Existence of reducible examples of genuinely partially hyperbolic actions in any odd dimension greater than 9. Reducible examples of  $\mathbb{Z}^2$  actions on torus  $\mathbb{T}^N$  viewed as  $\mathbb{T}^{n_1} \times \mathbb{T}^{n_2}$  ( $N = n_1 + n_2$ ) are obtained from generator  $g_i$  of  $\mathbb{Z}^2$ acting by  $A_{i,j}$  on  $\mathbb{T}^{n_j}$ , where  $i, j \in \{1, 2\}$  and  $A_{i,j}$  are matrices in  $GL(n_j, \mathbb{Z})$ . If such an action is genuinely partially hyperbolic then one of the reduced actions (on tori of dimension  $n_1$  or  $n_2$ ) would have to be genuinely partially hyperbolic. This, along with the preceding discussion on dimensions less than 5 implies that there are no reducible examples on the torus of dimension N = 7.

In any odd dimension  $N \ge 9$ , as proven in Section 9.1, there are no irreducible examples. However reducible examples exist as products of purely hyperbolic actions (which exist already in dimension 3) and the irreducible examples previously produced in Section 9.4. For example, on the nine-dimensional torus, in the construction above take  $n_1 = 3$ ,  $n_2 = 6$ , let  $A_{1,2}$  and  $A_{2,2}$  be commuting matrices from the Section 9.3 and for  $A_{1,1}$  and  $A_{2,1}$  choose any two  $3 \times 3$  commuting hyperbolic integer matrices which generate an irreducible  $\mathbb{Z}^2$  action (for various examples see [19]).

## References

- E. J. BENVENISTE, Rigidity of isometric lattice actions on compact Riemannian manifolds, *Geom. Funct. Anal.* 10 (2000), 516–542. MR 2001j:37055 Zbl 0970.22012
- [2] P. E. BLANKSBY and H. L. MONTGOMERY, Algebraic integers near the unit circle, *Acta Arith.* 18 (1971), 355–369. MR 45 #5082 Zbl 0221.12003
- [3] C. CAMACHO and A. LINS NETO, *Geometric Theory of Foliations*, Birkhäuser, Boston, MA, 1985. MR 87a:57029 Zbl 0568.57002
- [4] D. DAMJANOVIĆ, Central extensions of simple Lie groups and rigidity of some abelian partially hyperbolic algebraic actions, J. Mod. Dyn. 1 (2007), 665–688. MR 2008k:37062 Zbl 1139.37017
- [5] D. DAMJANOVIĆ and A. KATOK, Local rigidity of partially hyperbolic actions. II. The geometric method and restrictions of Weyl chamber flows on  $SL(n, \mathbb{R})/\Gamma$ , preprint.
- [6] \_\_\_\_\_, Local rigidity of actions of higher rank abelian groups and KAM method, *Electron. Res. Announc. Amer. Math. Soc.* **10** (2004), 142–154. MR 2005k:37044 Zbl 1068.37015
- [7] \_\_\_\_\_, Periodic cycle functionals and cocycle rigidity for certain partially hyperbolic  $\mathbb{R}^k$  actions, *Discrete Contin. Dyn. Syst.* **13** (2005), 985–1005. MR 2006g:37046 Zbl 1109.37029
- [8] \_\_\_\_\_, Local rigidity of restrictions of Weyl chamber flows, C. R. Math. Acad. Sci. Paris 344 (2007), 503–508. MR 2008k:37061 Zbl 1113.37015
- [9] \_\_\_\_\_, Local rigidity of certain homogeneous parabolic actions I: A model case, 2010, preprint.
- [10] R. DE LA LLAVE, A tutorial on KAM theory, in *Smooth Ergodic Theory and its Applications* (Seattle, WA, 1999), *Proc. Sympos. Pure Math.* 69, Amer. Math. Soc., Providence, RI, 2001, pp. 175–292. MR 2002h:37123 Zbl 1055.37064
- [11] M. EINSIEDLER and T. FISHER, Differentiable rigidity for hyperbolic toral actions, *Israel J. Math.* 157 (2007), 347–377. MR 2342454 Zbl 1112.37022
- [12] D. FISHER, First cohomology and local rigidity of group actions, Ann. of Math., to appear.
- [13] \_\_\_\_\_, Local rigidity of group actions: past, present, future, in *Dynamics, Ergodic Theory, and Geometry, Math. Sci. Res. Inst. Publ.* 54, Cambridge Univ. Press, Cambridge, 2007, pp. 45–97. MR 2009g:37021 Zbl 1144.22008
- [14] D. FISHER and G. MARGULIS, Local rigidity of affine actions of higher rank groups and lattices, Ann. of Math. 170 (2009), 67–122. MR 2521112 Zbl 1186.22015
- [15] R. S. HAMILTON, The inverse function theorem of Nash and Moser, Bull. Amer. Math. Soc. 7 (1982), 65–222. MR 83j:58014 Zbl 0499.58003
- [16] M. W. HIRSCH, C. C. PUGH, and M. SHUB, *Invariant Manifolds, Lecture Notes in Math.* 583, Springer-Verlag, New York, 1977. MR 58 #18595 Zbl 0355.58009

- [17] S. HURDER, Affine Anosov actions, *Michigan Math. J.* 40 (1993), 561–575. MR 94g:58165
   Zbl 0809.54035
- [18] A. KATOK and S. KATOK, Higher cohomology for abelian groups of toral automorphisms. II. The partially hyperbolic case, and corrigendum, *Ergodic Theory Dynam. Systems* 25 (2005), 1909–1917. MR 2006g:37047 Zbl 1109.57023
- [19] A. KATOK, S. KATOK, and K. SCHMIDT, Rigidity of measurable structure for Z<sup>d</sup>-actions by automorphisms of a torus, *Comment. Math. Helv.* 77 (2002), 718–745. MR 2003h:37007 Zbl 1035.37005
- [20] A. KATOK and A. KONONENKO, Cocycles' stability for partially hyperbolic systems, *Math. Res. Lett.* 3 (1996), 191–210. MR 97d:58152 Zbl 0853.58082
- [21] A. KATOK and J. LEWIS, Local rigidity for certain groups of toral automorphisms, *Israel J. Math.* **75** (1991), 203–241. MR 93g:58076 Zbl 0785.22012
- [22] \_\_\_\_\_, Global rigidity results for lattice actions on tori and new examples of volume-preserving actions, *Israel J. Math.* **93** (1996), 253–280. MR 96k:22021 Zbl 0857.57038
- [23] A. KATOK and R. J. SPATZIER, First cohomology of Anosov actions of higher rank abelian groups and applications to rigidity, *Inst. Hautes Études Sci. Publ. Math.* (1994), 131–156. MR 96c:58132 Zbl 0819.58027
- [24] \_\_\_\_\_, Subelliptic estimates of polynomial differential operators and applications to rigidity of abelian actions, *Math. Res. Lett.* **1** (1994), 193–202. MR 95b:35042 Zbl 0836.57026
- [25] \_\_\_\_\_, Differential rigidity of Anosov actions of higher rank abelian groups and algebraic lattice actions, *Tr. Mat. Inst. Steklova* **216** (1997), 292–319. MR 99i:58118 Zbl 0938.37010
- [26] Y. KATZNELSON, Ergodic automorphisms of T<sup>n</sup> are Bernoulli shifts, Israel J. Math. 10 (1971), 186–195. MR 45 #3672 Zbl 0219.28014
- [27] L. KRONECKER, Zwei Sätze über Gleichungen mit ganzzahligen Coefficienten, J. Reine Angew. Math. 53 (1857).
- [28] V. F. LAZUTKIN, KAM Theory and Semiclassical Approximations to Eigenfunctions, Ergeb. Math. Grenzgeb. 24, Springer-Verlag, New York, 1993. MR 94m:58069 Zbl 0814.58001
- [29] G. A. MARGULIS and N. QIAN, Rigidity of weakly hyperbolic actions of higher real rank semisimple Lie groups and their lattices, *Ergodic Theory Dynam. Systems* 21 (2001), 121–164. MR 2003a:22019 Zbl 0976.22009
- [30] D. MIECZKOWSKI, The first cohomology of parabolic actions for some higher-rank abelian groups and representation theory, *J. Mod. Dyn.* 1 (2007), 61–92. MR 2007i:22015 Zbl 1119. 22009
- [31] J. MOSER, On commuting circle mappings and simultaneous Diophantine approximations, *Math. Z.* 205 (1990), 105–121. MR 92e:58120 Zbl 0689.58031
- [32] Y. B. PESIN, Lectures on Partial Hyperbolicity and Stable Ergodicity, Zur. Lect. Adv. Math., European Mathematical Society (EMS), Zürich, 2004. MR 2005j:37039 Zbl 1098.37024
- [33] F. RODRIGUEZ HERTZ, Stable ergodicity of certain linear automorphisms of the torus, Ann. of Math. 162 (2005), 65–107. MR 2007d:37043 Zbl 1098.37028
- [34] A. N. STARKOV, The first cohomology group, mixing, and minimal sets of the commutative group of algebraic actions on a torus, J. Math. Sci. (New York) 95 (1999), 2576–2582. MR 2000m:37005 Zbl 0984.37004
- [35] W. A. VEECH, Periodic points and invariant pseudomeasures for toral endomorphisms, *Ergodic Theory Dynam. Systems* 6 (1986), 449–473. MR 87m:58096 Zbl 0616.28009
- [36] Z. J. WANG, Local rigidity of partially hyperbolic actions, *J. Mod. Dyn.* 4 (2010), 271–327. Zbl pre05790308

- [37] Z. J. WANG, New cases of differentiable rigidity for partially hyperbolic actions: symplectic groups and resonance directions, *J. Mod. Dyn.*, to appear.
- [38] E. ZEHNDER, Generalized implicit function theorems with applications to some small divisor problems. I, Comm. Pure Appl. Math. 28 (1975), 91–140. MR 52 #1764 Zbl 0309.58006

(Received January 31, 2007) (Revised September 22, 2008)

E-mail address: dani@rice.edu

DEPARTMENT OF MATHEMATICS, RICE UNIVERSITY, 6100 MAIN STREET, HOUSTON, TX 77005, UNITED STATES

*E-mail address*: katok\_a@math.psu.edu

DEPARTMENT OF MATHEMATICS, THE PENNSYLVANIA STATE UNIVERSITY, UNIVERSITY PARK, STATE COLLEGE, 16802, UNITED STATES

http://www.math.psu.edu/katok\_a/

### ISSN 0003-486X

#### ANNALS OF MATHEMATICS

This periodical is published bimonthly by the Department of Mathematics at Princeton University with the cooperation of the Institute for Advanced Study. Annals is typeset in  $T_EX$  by Sarah R. Warren and produced by Mathematical Sciences Publishers. The six numbers each year are divided into two volumes of three numbers each.

## **Editorial correspondence**

Papers submitted for publication and editorial correspondence should be addressed to Maureen Schupsky, Annals of Mathematics, Fine Hall-Washington Road, Princeton University, Princeton, NJ, 08544-1000 U.S.A. The e-mail address is annals@math.princeton.edu.

#### Preparing and submitting papers

The Annals requests that all papers include an abstract of about 150 words which explains to the nonspecialist mathematician what the paper is about. It should not make any reference to the bibliography. Authors are encouraged to initially submit their papers electronically and in PDF format. Please send the file to: annals@math.princeton.edu or to the Mathematics e-print arXiv: front.math.ucdavis.edu/submissions. If a paper is submitted through the arXiv, then please e-mail us with the arXiv number of the paper.

#### Proofs

A PDF file of the galley proof will be sent to the corresponding author for correction. If requested, a paper copy will also be sent to the author.

## Offprints

Authors of single-authored papers will receive 30 offprints. (Authors of papers with one co-author will receive 15 offprints, and authors of papers with two or more co-authors will receive 10 offprints.) Extra offprints may be purchased through the editorial office.

### Subscriptions

The price for a print and online subscription, or an online-only subscription, is \$390 per year for institutions. In addition, there is a postage surcharge of \$40 for print subscriptions that are mailed to countries outside of the United States. Individuals interested in subscriptions for their own personal use should contact the publisher at the address below. Subscriptions and changes of address should be sent to Mathematical Sciences Publishers, Department of Mathematics, University of California, Berkeley, CA 94720-3840 (e-mail: contact@mathscipub.org; phone: 1-510-643-8638; fax: 1-510-295-2608). (Checks should be made payable to "Mathematical Sciences Publishers".)

## Back issues and reprints

Orders for missing issues and back issues should be sent to Mathematical Sciences Publishers at the above address. Claims for missing issues must be made within 12 months of the publication date. Online versions of papers published five or more years ago are available through JSTOR (www.jstor.org).

#### Microfilm

Beginning with Volume 1, microfilm may be purchased from NA Publishing, Inc., 4750 Venture Drive, Suite 400, PO Box 998, Ann Arbor, MI 48106-0998; phone: 1-800-420-6272 or 734-302-6500; email: info@napubco.com, website: www.napubco.com/contact.html.

ALL RIGHTS RESERVED UNDER THE BERNE CONVENTION AND THE UNIVERSAL COPYRIGHT CONVENTION

Copyright © 2010 by Princeton University (Mathematics Department) Printed in U.S.A. by Sheridan Printing Company, Inc., Alpha, NJ

# TABLE OF CONTENTS

SHIGERU MUKAI. Curves and symmetric spaces, II	1539 - 15	558
MANJUL BHARGAVA. The density of discriminants of quintic rings and fields	1559–15	591
Shuji Saito and Kanetomo Sato. A finiteness theorem for zero-cycles over <i>p</i> -adic fields	1593–16	339
SYLVAIN CROVISIER. Birth of homoclinic intersections: a model for the central dynamics of partially hyperbolic systems	1641-16	377
JOSEPH BERNSTEIN and ANDRE REZNIKOV. Subconvexity bounds for triple <i>L</i> -functions and representation theory	1679–17	718
SÁNDOR J KOVÁCS and MAX LIEBLICH. Boundedness of families of canonically polarized manifolds: A higher dimensional analogue of Shafarevich's conjecture	1719–17	748
ANDREI TELEMAN. Instantons and curves on class VII surfaces		
DANIJELA DAMJANOVIĆ and ANATOLE KATOK. Local rigidity of partially hyperbolic actions I. KAM method and $\mathbb{Z}^k$ actions on the torus		858
JORGE LAURET. Einstein solvmanifolds are standard		
PASCAL COLLIN and HAROLD ROSENBERG. Construction of harmonic diffeomorphisms and minimal graphs	1879–19	906
JOHN LEWIS and KAJ NYSTRÖM. Boundary behavior and the Martin boundary problem for $p$ harmonic functions in Lipschitz domains	1907–19	948
JENS MARKLOF and ANDREAS STRÖMBERGSSON. The distribution of free path lengths in the periodic Lorentz gas and related lattice point problems		)33
ÉTIENNE FOUVRY and JÜRGEN KLÜNERS. On the negative Pell equation	2035–21	104
LEX G. OVERSTEEGEN and EDWARD D. TYMCHATYN. Extending isotopies of planar continua	2105-21	133
JAN BRUINIER and KEN ONO. Heegner divisors, <i>L</i> -functions and harmonic weak Maass forms	2135-21	181
FLORIAN POP. Henselian implies large	2183-21	195
MIKHAIL BELOLIPETSKY, TSACHIK GELANDER, ALEXANDER LUBOTZKY and ANER SHALEV. Counting arithmetic lattices and surfaces	2197-22	221
SASHA SODIN. The spectral edge of some random band matrices	2223-22	251